

Contents lists available at ScienceDirect

Expositiones Mathematicae

journal homepage: www.elsevier.de/exmath



Stochastic integrals for spde's: A comparison

Robert C. Dalang^{*,1}, Lluís Quer-Sardanyons²

^a Institut de Mathématiques, Ecole Polytechnique Fédérale, Station 8, CH-1015 Lausanne, Switzerland ^b Departament de Matemàtiques, Universitat Autònoma de Barcelona, 08193 Bellaterra (Barcelona), Spain

ARTICLE INFO

Article history: Received 21 December 2009 Received in revised form 9 June 2010

2000 Mathematics Subject Classification: primary 60H15 secondary 60H05

Keywords:

Stochastic integral Stochastic partial differential equation Martingale measure Cylindrical Wiener process Hilbert-space-valued Wiener process Spatially homogeneous Gaussian noise Stochastic heat equation Stochastic wave equation Random field solution

ABSTRACT

We present the Walsh theory of stochastic integrals with respect to martingale measures, and various extensions of this theory, alongside of the Da Prato and Zabczyk theory of stochastic integrals with respect to Hilbert-space-valued Wiener processes, and we explore the links between these theories. Somewhat surprisingly, the end results of both theories turn out to be essentially equivalent. We then show how each theory can be used to study stochastic partial differential equations, with an emphasis on the stochastic heat and wave equations driven by spatially homogeneous Gaussian noise that is white in time. We compare the solutions produced by the different theories.

© 2010 Elsevier GmbH. All rights reserved.

1. Introduction

The theory of stochastic partial differential equations (spde's) developed on the one hand, from the work of Walsh [41], and on the other hand, through work on stochastic evolution equations in Hilbert spaces, such as in [13]. Important milestones in the latter approach are the books of Da Prato and Zabczyk [14] and Rozovskii [37] (see also Krylov and Rozovskii [23] and Krylov [22]).

These two approaches led to the development of two distinct schools of study for spde's, based on different theories of stochastic integration: the Walsh theory, which emphasizes integration with respect to worthy martingale measures, and a theory of integration with respect to Hilbert-spacevalued processes, as expounded in [14]. A consequence of the presence of these separate theories is

0723-0869/\$ – see front matter 0 2010 Elsevier GmbH. All rights reserved. doi:10.1016/j.exmath.2010.09.005

^{*} Corresponding author. Tel.: +41 21 6932551; fax: +41 21 6935380.

E-mail addresses: robert.dalang@epfl.ch (R.C. Dalang), quer@mat.uab.cat (L. Quer-Sardanyons).

¹ Partly supported by the Swiss National Foundation for Scientific Research.

² Partly supported by the grant MCI-FEDER Ref. MTM2009-08869 from the Ministerio de Ciencia e Innovación, Spain.

that the literature published by each of the two schools is difficult to access when one has been trained in the other school. This is unfortunate since both approaches have advantages and in some problems, using both approaches can be useful (one example of this is [10]).

The objective of this paper is to help create links between these two schools of study. It is addressed to researchers who have some familiarity with at least one of the two approaches. We develop both theories, and explore the links between the two. Then we show how each theory is used to study spde's. The Walsh theory emphasizes solutions that are random fields, while [14] centers around solutions in Hilbert spaces of functions. Each theory is presented rather succinctly, the main focus being on relationships between the theories. We show that these theories often (but not always) lead to the same solutions to various spde's.

It should be mentioned that the general theory of integration with respect to Hilbert-spacevalued processes and its generalizations – such as the stochastic integral with respect to cylindrical processes – was well-developed several years before [41] and more than a decade before reference [14] appeared: see, for instance, the book of Métivier and Pellaumail [25]. This reference, and several others, are cited in [14] and [41]. However, Walsh preferred to develop his own integral, even though he realized that the two were related (see the Notes at the end of [41]).

Here, we present in Section 2.1 a modern formulation of the theory of stochastic integrals with respect to cylindrical Wiener processes, as developed in [25], as a unifying integral behind most of those that were introduced later on. This integral is briefly recalled in Section 2.1. In Section 2.2, we show how spatially homogeneous Gaussian noise that is white in time can be viewed as a cylindrical Wiener process on a particular Hilbert space. Emphasizing this type of noise is natural, since in recent vears, following in particular the papers of Mueller [28], Dalang and Frangos [7], Dalang [6] and Peszat and Zabczyk [31,32], this type of noise has been used by several researchers. This is due in part to the fact that it leads to a theory of non-linear spde's in spatial dimensions greater than 1, while nonlinear spde's driven by space-time white noise generally only have a solution in spatial dimension 1. In Section 2.3, we show (Proposition 2.6) that the Walsh stochastic integral and the extension presented by Dalang [6] and Nualart and Quer-Sardanyons [29] can be viewed as integrals as defined in Section 2.1. Section 2.4 gives a wide class of integrable processes. In Section 2.5, we discuss the relationship between this integral and the function-valued stochastic integral introduced by Dalang and Mueller in [9]. A further extension of real-valued integrals to Hilbert-space-valued stochastic integrals was developed in [29,35,38]; these extensions were motivated by the needs of Malliavin calculus: indeed, the so-called Malliavin derivative of the solution to an spde satisfies a stochastic integral equation which requires a Hilbert-space-valued integral. We give a unified presentation of these extensions in Section 2.6.

In Section 3, we sketch the construction of the infinite dimensional stochastic integral in the setup of Da Prato and Zabczyk [14]. We also make use of the more recent presentation of Prévôt and Röckner [33]. In Section 3.1, we recall some basic properties of Hilbert–Schmidt operators. Section 3.2 gives the relationship between a Hilbert-space-valued Wiener process and a cylindrical Brownian motion, in the case where the covariance operator has finite trace. Hilbert-space-valued stochastic integrals are defined in Section 3.3. In particular, we show in Proposition 3.4 how this infinite-dimensional stochastic integral can be written as a series of Itô stochastic integrals. This is used in Section 3.4 to show how the integrals of Section 2 can be interpreted in the infinite-dimensional context. The case of covariance operators with infinite-trace is discussed in Section 3.5. We do not discuss Banach-space-valued stochastic integrals, for which we refer to [2,40]. Finally, in Section 3.6, we establish the somewhat unexpected but interesting fact that the extension of the Walsh stochastic integral presented in Section 2.6 and the Da Prato and Zabczyk integral of Section 3.5 are in fact equivalent.

It is well-known that in certain cases, the Hilbert-space-valued integral is equivalent to a martingale-measure stochastic integral. For instance, it is pointed out in [14, Section 4.3] that when the random perturbation is space-time white noise, then Walsh's stochastic integral in [41] is equivalent to an infinite-dimensional stochastic integral as in [14] (see also [19]). Of course, space-time white noise is only a special case of spatially homogeneous noise, and we are interested in comparing solutions to spde's driven by this more general noise. The function-valued approach of [9] gives solutions to spde's for which it is not known if a random field solution exists, and the Hilbert-space approach is even more general. However, for a wide class of spde's that have solutions in two or more

of these formulations, such as the stochastic heat equation ($d \ge 1$) and wave equation ($d \in \{1, 2, 3\}$) driven by spatially homogeneous noise, we will show that the solutions turn out to be equivalent. One does not expect this to be the case in all situations. Indeed, there are a few cases in which a solution exists with one approach and is known *not to exist* in one of the others. For instance, for noise concentrated on a hyperplane, as considered in [8], the authors establish existence of function-valued solutions and show that there is no random field solution.

In Section 4, we consider spde's driven by spatially homogeneous noise, with an emphasis on the stochastic heat and wave equations. In Sections 4.1–4.3, we discuss the random field approach, and we use the stochastic integral of Section 2.3 to extend the result of [6] to arbitrary initial conditions (Theorem 4.3). In Section 4.4, we discuss the Hilbert-space-valued approach to the study of the same equations, using the approach of [32]. In Section 4.5, we show that the mild random field solution of Theorem 4.3, when interpreted as a Hilbert-space-valued process, yields the solution given in [32]. This is achieved by identifying the multiplicative non-linearity with an appropriate Hilbert–Schmidt operator, and using the relationships between stochastic integrals identified in Section 3. Since the two solutions are defined using different Hilbert spaces, the embedding from one Hilbert space to the other has to be written explicitly. Finally, in Section 4.6, we compare the random field solution of the stochastic wave equation with the function-valued solution constructed in [9]. Again, in cases where both types of solutions are defined, that is, in spatial dimensions $d \in \{1, 2, 3\}$, we show that the random field solution yields the function-valued solution (Theorem 4.13). Overall, Section 4 unifies the existing literature on the stochastic heat and wave equations driven by spatially homogeneous noise, and clarifies the relationships between the various approaches.

2. Stochastic integrals with respect to a spatially homogeneous Gaussian noise

In this section, we recall in Section 2.1 the notion of cylindrical Wiener process and the stochastic integral with respect to such processes. In Section 2.2, we introduce a spatially homogeneous Gaussian noise that is white in time, and we show how to interpret this noise as a cylindrical Wiener process. Building on material presented in [29], we relate in Section 2.3 the stochastic integral with respect to this particular cylindrical Wiener process with Walsh's martingale measure stochastic integral and the extension given by Dalang in [6]. Some examples of integrands are given in Section 2.4. In Section 2.5, we discuss the function-valued extension given in Dalang and Mueller [9]. Finally, in Section 2.6, we give a unified presentation of the Hilbert-space-valued stochastic integral developed in [29,35,38].

2.1. Stochastic integration with respect to a cylindrical Wiener process

Fix a separable Hilbert space V with inner product $\langle \cdot, \cdot \rangle_V$. Following [18,25], we define the general notion of cylindrical Wiener process in V.

Definition 2.1. Let *Q* be a symmetric (self-adjoint) and non-negative definite bounded linear operator on *V*. A family of random variables $B = \{B_t(h), t \ge 0, h \in V\}$ is a *cylindrical Wiener process* on *V* if the following two conditions are fulfilled:

1. for any $h \in V$, { $B_t(h)$, $t \ge 0$ } defines a Brownian motion with variance $t \langle Qh, h \rangle_V$; 2. for all $s, t \in \mathbb{R}_+$ and $h, g \in V$,

 $E(B_s(h)B_t(g)) = (s \wedge t) \langle Qh, g \rangle_V,$

where $s \wedge t := \min(s, t)$. If $Q = Id_V$ is the identity operator in V, then B will be called a *standard cylindrical Wiener process*. We will refer to Q as the *covariance* of B.

Let \mathcal{F}_t be the σ -field generated by the random variables { $B_s(h)$, $h \in V$, $0 \le s \le t$ } and the *P*-null sets. We define the predictable σ -field as the σ -field in $[0, T] \times \Omega$ generated by the sets { $(s, t] \times A, A \in \mathcal{F}_s, 0 \le s < t \le T$ }.

We denote by V_Q (the completion of) the Hilbert space V endowed with the inner-product

$$\langle h, g \rangle_{V_0} := \langle Qh, g \rangle_V, \qquad h, g \in V.$$

We can now define the stochastic integral of any predictable square-integrable process with values in V_Q , as follows. Let $(v_j)_j$ be a complete orthonormal basis of the Hilbert space V_Q . For any predictable process $g \in L^2(\Omega \times [0, T]; V_Q)$, it turns out that the following series is convergent in $L^2(\Omega, \mathcal{F}, P)$ and the sum does not depend on the chosen orthonormal system:

$$g \cdot B := \sum_{j=1}^{\infty} \int_0^T \langle g_s, v_j \rangle_{V_Q} \, dB_s(v_j).$$
(2.1)

We notice that each summand in the above series is a classical Itô integral with respect to a standard Brownian motion, and the resulting stochastic integral is a real-valued random variable. The stochastic integral $g \cdot B$ is also denoted by $\int_0^T g_s dB_s$. The independence of the terms in the series (2.1) leads to the isometry property

$$E\left(\left(g \cdot B\right)^{2}\right) = E\left(\left(\int_{0}^{T} g_{s} dB_{s}\right)^{2}\right) = E\left(\int_{0}^{T} \|g_{s}\|_{V_{Q}}^{2} ds\right).$$

We note that there is an alternative way of defining this integral: one can start by defining the stochastic integral in (2.1) for a class of *simple* predictable V_Q -valued processes, and then use the isometry property to extend the integral to elements of $L^2(\Omega \times [0, T]; V_Q)$ by checking that these simple processes are dense in this set.

2.2. Spatially homogeneous noise as a cylindrical Wiener process

We now define the Gaussian random noise that will play a central role in this paper. On a complete probability space (Ω, \mathcal{F}, P) , we consider a family of mean zero Gaussian random variables $W = \{W(\varphi), \varphi \in C_0^{\infty}(\mathbb{R}^{d+1})\}$, where $C_0^{\infty}(\mathbb{R}^{d+1})$ denotes the space of infinitely differentiable functions with compact support, with covariance

$$E(W(\varphi)W(\psi)) = \int_0^\infty dt \, \int_{\mathbb{R}^d} \Lambda(dx) \, (\varphi(t) * \tilde{\psi}(t))(x), \tag{2.2}$$

where "*" denotes convolution in the spatial variable and $\tilde{\psi}(t, x) := \psi(t, -x)$.

In the above, Λ is a non-negative and non-negative definite tempered measure on \mathbb{R}^d , it is therefore the Fourier transform of a non-negative tempered measure μ on \mathbb{R}^d . That is, by definition of the Fourier transform on the space $S'(\mathbb{R}^d)$ of tempered distributions (see [39]), for all φ belonging to the space $S(\mathbb{R}^d)$ of rapidly decreasing \mathcal{C}^{∞} functions,

$$\int_{\mathbb{R}^d} \varphi(x) \Lambda(dx) = \int_{\mathbb{R}^d} \mathcal{F}\varphi(\xi) \, \mu(d\xi),$$

and there is an integer $m \ge 1$ such that

$$\int_{\mathbb{R}^d} (1+|\xi|^2)^{-m} \mu(d\xi) < \infty.$$
(2.3)

We have denoted by $\mathcal{F}\varphi$ the Fourier transform of $\varphi \in \mathcal{S}(\mathbb{R}^d)$:

$$\mathcal{F}\varphi(\xi) = \int_{\mathbb{R}^d} \varphi(x) e^{-2\pi i \xi \cdot x} \, dx.$$

The measure μ is called the *spectral measure* of W and is necessarily symmetric (see [39, Chap. VII, Théorème XVII]). The covariance (2.2) can also be written, using elementary properties of the Fourier

transform, as

$$E(W(\varphi)W(\psi)) = \int_0^\infty dt \, \int_{\mathbb{R}^d} \mu(d\xi) \, \mathcal{F}\varphi(t)(\xi) \, \overline{\mathcal{F}\psi(t)(\xi)}.$$

Remark 2.2. In the case where the measure $\Lambda(dx)$ is absolutely continuous with respect to Lebesgue measure on \mathbb{R}^d , with density *f*, formula (2.2) becomes

$$\int_0^\infty dt \, \int_{\mathbb{R}^d} dx \int_{\mathbb{R}^d} dy \, \varphi(t,x) f(x-y) \, \psi(t,y),$$

which makes clear the spatially homogeneous character of the noise.

It is natural to associate a Hilbert space with W: let U the completion of the Schwartz space $\mathcal{S}(\mathbb{R}^d)$ endowed with the semi-inner product

$$\langle \varphi, \psi \rangle_{U} = \int_{\mathbb{R}^{d}} \Lambda(dx)(\varphi * \tilde{\psi})(x) = \int_{\mathbb{R}^{d}} \mu(d\xi) \,\mathcal{F}\varphi(\xi) \,\overline{\mathcal{F}\psi(\xi)},\tag{2.4}$$

 $\varphi, \psi \in \mathcal{S}(\mathbb{R}^d)$, and associated semi-norm $\|\cdot\|_U$. Then U is a separable Hilbert space that may contain Schwartz distributions (see [6, Example 6]).

Remark 2.3. Let $\tilde{L}^2(\mathbb{R}^d, d\mu)$ be the subspace of $L^2(\mathbb{R}^d, d\mu)$ consisting of functions ϕ such that $\tilde{\phi} = \phi$. It is not difficult to check that one can identify U with the set $\{\Psi \in S'(\mathbb{R}^d) : \Psi = \mathcal{F}^{-1}\phi$, where $\phi \in \mathcal{F}^{-1}(\mathbb{R}^d)$ $\tilde{L}^2(\mathbb{R}^d, d\mu)$, with inner product

$$\langle \mathcal{F}^{-1}\phi, \mathcal{F}^{-1}\varphi \rangle_U = \langle \phi, \varphi \rangle_{L^2(\mathbb{R}^d, d\mu)}, \qquad \phi, \varphi \in \tilde{L}^2(\mathbb{R}^d, d\mu).$$

We fix a time interval [0, T] and we set $U_T := L^2([0, T]; U)$. This set is equipped with the norm given by

$$||g||_{U_T}^2 = \int_0^T ||g(s)||_U^2 ds.$$

We now associate a cylindrical Wiener process to W, as follows. A direct calculation using (2.2) shows that the generalized Gaussian random field $\{W(\varphi), \varphi \in C_0^{\infty}([0, T] \times \mathbb{R}^d)\}$ is a random linear functional, in the sense that $W(a\varphi + b\psi) = aW(\varphi) + bW(\psi)$, and $\varphi \mapsto W(\varphi)$ is an isometry from $(C_0^{\infty}([0, T] \times \mathbb{R}^d), \|\cdot\|_{U_T})$ into $L^2(\Omega, \mathcal{F}, P)$. The following lemma identifies the completion of $\mathcal{C}_{0}^{\infty}([0, T] \times \mathbb{R}^{d})$ with respect to $\|\cdot\|_{U_{T}}$.

Lemma 2.4. The space $\mathcal{C}_0^{\infty}([0,T] \times \mathbb{R}^d)$ is dense in $U_T = L^2([0,T];U)$ for $\|\cdot\|_{U_T}$.

Proof. Following [29], we will use the notation $\varphi_1(\cdot)$ to indicate that φ_1 is a function $t \mapsto \varphi_1(t)$ of the time-variable, and $\varphi_2(\star)$ to indicate that φ_2 is a function $x \mapsto \varphi_2(x)$ of the spatial variable. Let *C* denote the closure of $\mathcal{C}_0^{\infty}([0, T] \times \mathbb{R}^d)$ in U_T for $\|\cdot\|_{U_T}$. Clearly, *C* is a subspace of U_T . The

proof can be split into three parts.

Step 1. We show that elements of U_T of the form $\varphi_1(\cdot)\varphi_2(\star)$, where $\varphi_1 \in C_0^{\infty}(\mathbb{R}_+;\mathbb{R})$ with support included in [0, T] and $\varphi_2 \in \mathcal{S}(\mathbb{R}^d)$, belong to *C*. Using the fact that

$$\int_{\mathbb{R}^d} \Lambda(dx) \, (|\varphi_2| * |\tilde{\varphi}_2|)(x) < \infty$$

because Λ is a tempered measure and $|\varphi_2| * |\tilde{\varphi}_2|$ decreases rapidly, together with dominated convergence, one checks that there is a sequence $(\varphi_2^n)_n \subset C_0^{\infty}(\mathbb{R}^d)$ such that $\lim_{n\to\infty} \|\varphi_2 - \varphi_2^n\|_{H} = 0$. Then, by the very definition of the norm in U_T , one easily proves that $\lim_{n\to\infty} \|\varphi_1\varphi_2 - \varphi_1\varphi_2^n\|_{U_T} = 0$. Therefore, $\varphi_1(\cdot)\varphi_2(\star) \in U_T$.

Step 2. Suppose that we are given $\varphi_1 \in L^2([0, T]; \mathbb{R})$ and $\varphi_2 \in \mathcal{S}(\mathbb{R}^d)$. We show that $\varphi_1(\cdot)\varphi_2(\star) \in C$. Indeed, let $(\varphi_1^n)_n \in C_0^{\infty}(\mathbb{R}_+)$ be such that, for all *n*, the support of φ_1^n is contained in [0, T] and $\varphi_1^n \to \varphi_1$ in $L^2([0, T]; \mathbb{R})$. Then $\varphi_1^n \varphi_2 \in C$ by Step 1, and one checks that $\varphi_1^n \varphi_2$ converges, as *n* tends to infinity, to $\varphi_1 \varphi_2$ in U_T . Therefore, $\varphi_1(\cdot)\varphi_2(\star) \in C$.

Step 3. Suppose that $\varphi \in U_T$. We show that $\varphi \in C$. Indeed, let $(e_j)_j$ be a complete orthonormal basis of U with $e_j \in S(\mathbb{R}^d)$, for all j. Then, since $\varphi(s) \in U$ for any $s \in [0, T]$,

$$\|\varphi\|_{U_T}^2 = \int_0^T \|\varphi(s)\|_U^2 \, ds = \sum_{j=1}^\infty \int_0^T \langle \varphi(s), e_j \rangle_U^2 \, ds.$$

In particular, for any $j \ge 1$, the function $s \mapsto \langle \varphi(s), e_j \rangle_U$ belongs to $L^2([0, T]; \mathbb{R})$. Thus, it follows from Step 2 that

$$\varphi^{n}(\cdot) := \sum_{j=1}^{n} \langle \varphi(\cdot), e_{j} \rangle_{U} e_{j}$$

belongs to *C*. Moreover, it is straightforward to verify that $\|\varphi - \varphi^n\|_{U_T}^2 \to 0$ as $n \to \infty$. This shows that $\varphi \in C$. \Box

Therefore, taking into account the above lemma, $W(\varphi)$ can be defined for all $\varphi \in U_T$ following the standard method for extending an isometry. This establishes the following property.

Proposition 2.5. For $t \ge 0$ and $\varphi \in U$, set $W_t(\varphi) = W(1_{[0,t]}(\cdot)\varphi(\star))$. Then the process $W = \{W_t(\varphi), t \ge 0, \varphi \in U\}$ is a cylindrical Wiener process as defined in Section 2.1, with V there replaced by U and Q = Id_U. In particular, for any $\varphi \in U$, $\{W_t(\varphi), t \ge 0\}$ is a Brownian motion with variance $t ||\varphi||_U$ and for all s, $t \ge 0$ and $\varphi, \psi \in U$, $E(W_t(\varphi)W_s(\psi)) = (s \land t)\langle\varphi,\psi\rangle_U$.

With this proposition, it becomes possible to use the stochastic integral defined in Section 2.1. This defines the stochastic integral $g \cdot W$ for all $g \in L^2(\Omega \times [0, T]; U) \equiv L^2(\Omega; U_T)$. By definition of U, the complete orthonormal basis $(e_i)_i$ in the definition of $g \cdot W$ can be chosen such that $(e_i)_i \subset S(\mathbb{R}^d)$.

Before discussing this further, we first relate the statement of Proposition 2.5 to Walsh's theory of stochastic integrals with respect to martingale measures. Let us recall that Walsh's theory of stochastic integration is based on the concept of *martingale measure*, which is a stochastic process of the form $\{M_t(A), \mathcal{F}_t, t \in [0, T], A \in \mathcal{B}_b(\mathbb{R}^d)\}$, where $\mathcal{B}_b(\mathbb{R}^d)$ denotes the set of bounded Borel sets of \mathbb{R}^d , and $(\mathcal{F}_t)_t$ is a filtration satisfying the *usual conditions*. For the precise definition of a martingale measure, we refer to [41, Chapter 2]. Hence, in order to use Walsh's construction, one has first to extend the generalized random field $\{W(\varphi), \varphi \in \mathcal{C}_0^\infty(\mathbb{R}_+ \times \mathbb{R}^d)\}$ to a martingale measure. More precisely, using an approximation procedure similar to the one used in Lemma 2.4, one extends the definition of *W* to indicator functions of bounded Borel sets in $\mathbb{R}_+ \times \mathbb{R}^d$ (for details see [7] or [34, p. 13]). Then one sets

$$M_t(A) = W(1_{[0,t]}(\cdot)1_A(\star)), \qquad t \in [0,T], A \in \mathcal{B}_b(\mathbb{R}^d).$$
(2.5)

Moreover, if we let $(\mathcal{F}_t)_t$ be the filtration generated by $\{M_t(A), A \in \mathcal{B}_b(\mathbb{R}^d)\}$ (completed and made right-continuous), then the process $\{M_t(A), \mathcal{F}_t, t \in [0, T], A \in \mathcal{B}_b(\mathbb{R}^d)\}$ defines a worthy martingale measure in the sense of Walsh [41]. Its *covariance measure* is determined by

$$\langle M(A), M(B) \rangle_t = t \int_{\mathbb{R}^d} \Lambda(dx) (1_A * \tilde{1}_B)(x),$$

 $t \in [0, T]$, $A, B \in \mathcal{B}_b(\mathbb{R}^d)$, and its *dominating measure* coincides with the covariance measure (see [7]).

One easily checks that, for $\varphi \in \mathcal{S}(\mathbb{R}^d)$,

$$W_t(\varphi) = \int_{\mathbb{R}_+} \int_{\mathbb{R}^d} \mathbb{1}_{[0,t]}(s) \,\varphi(x) \, M(ds, dx),$$

where the integral on the right-hand side is Walsh's stochastic integral.

2.3. The real-valued stochastic integral for spatially homogeneous noise

The aim of this section is to exhibit the relationship between the stochastic integral constructed in Section 2.1 and the random field approach of Walsh [41] and Dalang [6]. Recall that the stochastic integral with respect to *M* defined in [41] only allows function-valued integrands, and this theory was extended in [6] in order to cover more general integrands, such as certain processes with values in the space of (Schwartz) distributions. We are going to show that these two integrals can be interpreted in the context of Section 2.1.

Recall that Walsh's stochastic integral $g \cdot M$ is defined when $g \in \mathcal{P}_+$, where \mathcal{P}_+ is the set of predictable processes $(\omega, t, x) \mapsto g(t, x; \omega)$ such that

$$\|g\|_{+}^{2}:=E\left(\int_{0}^{T}dt\int_{\mathbb{R}^{d}}\Lambda(dx)\left(|g(t,\star)|*|\tilde{g}(t,\star)|\right)<\infty\right)<\infty.$$

For $g \in \mathcal{P}_+$, we can consider that $g \in L^2(\Omega; U_T)$ and set

$$\|g\|_{0}^{2} := E(\|g\|_{U_{T}}^{2}) = E\left(\int_{0}^{T} dt \int_{\mathbb{R}^{d}} \Lambda(dx) \left(g(t, \star) * \tilde{g}(t, \star)\right)(x)\right).$$
(2.6)

In [6], Dalang considered (in the case $\Lambda(dx) = f(x)dx$) the set \mathcal{P}_0 , which is the completion with respect to $\|\cdot\|_0$ of the subset \mathcal{E}_0 of \mathcal{P}_+ that consists of functions $g(s, x; \omega)$ such that $x \mapsto g(s, x; \omega) \in \mathcal{S}(\mathbb{R}^d)$, for all s and ω , and he defined the stochastic integral $g \cdot M$ for all $g \in \mathcal{P}_0$.

Finally, in order to use the stochastic integral of Section 2.1, let $(e_j)_j \subset S(\mathbb{R}^d)$ be a complete orthonormal basis of U, and consider the cylindrical Wiener process $\{W_t(\varphi)\}$ defined in Proposition 2.5. For any predictable process $g \in L^2(\Omega \times [0, T]; U)$, the stochastic integral of g with respect to W is

$$g \cdot W = \int_0^T g_s \, dW_s := \sum_{j=1}^\infty \int_0^T \left\langle g_s, e_j \right\rangle_U \, dW_s(e_j), \tag{2.7}$$

and the isometry property is given by

$$E\left((g \cdot W)^2\right) = E\left(\left(\int_0^T g_s \, dW_s\right)^2\right) = E\left(\int_0^T \|g_s\|_U^2 \, ds\right). \tag{2.8}$$

We note that the right-hand side of (2.7) is essentially the definition of $W(\varphi)$ in [24,26,27]. We also use the notation

$$\int_0^T \int_{\mathbb{R}^d} g(s, y) \, W(ds, dy)$$

instead of $\int_0^T g_s dW_s$.

Proposition 2.6. (a) If $g \in \mathcal{P}_+$, then $g \in L^2(\Omega \times [0, T]; U)$ and $g \cdot M = g \cdot W$, where the left-hand side is a Walsh integral and the right-hand side is defined as in (2.7).

(b) If $g \in \mathcal{P}_0$, then $g \in L^2(\Omega \times [0, T]; U)$ and $g \cdot M = g \cdot W$, where the left-hand side is a Dalang integral and the right-hand side is defined as in (2.7).

Proof. Let us prove part (a) in the statement. We recall the inclusion $\mathcal{P}_+ \subset \mathcal{P}_0$, observed in [6], and we note that if $g \in \mathcal{P}_+$, then

$$\|g\|_{L^{2}(\Omega\times[0,T];U)}^{2} = E\left(\int_{0}^{T} dt \int_{\mathbb{R}^{d}} \Lambda(dx) \left(g(t,\star) * \tilde{g}(t,\star)\right)(x)\right) \le \|g\|_{+}^{2} < +\infty,$$
(2.9)

and, in particular $g \in L^2(\Omega \times [0, T]; U)$. Indeed, the equality in (2.9) holds when $g \in C_0^{\infty}([0, T] \times \mathbb{R}^d)$, and it also holds (by monotone approximation) for $g(s, x) = 1_{[a,b]}(s) 1_A(x)$, when A is a product of open intervals, and finally, it holds for all $g \in \mathcal{P}_+$ via the Monotone Class Theorem (using arguments similar to those in the proof of [4, Theorem 2.6(b)]).

Secondly, in order to check the equality of the integrals, we use the fact that the set of elementary processes is dense in $(\mathcal{P}_+, \|\cdot\|_+)$ (see [41, Proposition 2.3]). Hence, by inequality (2.9), it suffices to show that both integrals coincide when *g* is an elementary process of the form

$$g(t, x; \omega) = 1_{(a, b]}(t) 1_A(x) X(\omega),$$
(2.10)

where $0 \le a < b \le T$, $A \in \mathcal{B}_b(\mathbb{R}^d)$ and X is a bounded and \mathcal{F}_a -measurable random variable.

On one hand, when g has the particular form (2.10), according to [41] and (2.5),

$$\int_0^T \int_{\mathbb{R}^d} g(t, x) M(dt, dx) = [M_b(A) - M_a(A)] X = \left[W(1_{(0, b]}(\cdot) 1_A(\star)) - W(1_{(0, a]}(\cdot) 1_A(\star)) \right] X$$
$$= W(1_{(a, b]}(\cdot) 1_A(\star)) X.$$

On the other hand, by the very definition of the integral (2.7),

$$\int_0^T g_t \, dW_t = \sum_{j=1}^\infty \int_a^b X \langle 1_A, e_j \rangle_U \, dW_t(e_j) = X \sum_{j=1}^\infty \langle 1_A, e_j \rangle_U \left[W_b(e_j) - W_a(e_j) \right]$$
$$= X \sum_{j=1}^\infty \langle 1_A, e_j \rangle_U \, W(1_{(a,b]}(\cdot)e_j) = XW(1_{(a,b]}(\cdot)1_A(\star)),$$

which implies that

$$\int_0^T \int_{\mathbb{R}^d} g(t, x) M(dt, dx) = \int_0^T g_t \, dW_t,$$

for all g of the form (2.10). This concludes the first part of the proof.

Concerning part (b), let us point out that \mathcal{P}_0 is the completion of \mathcal{E}_0 with respect to $\|\cdot\|_0$ (see (2.6)), where the latter coincides with the norm in $L^2(\Omega \times [0, T]; U)$ for smooth elements. Hence, since $\mathcal{E}_0 \subset \mathcal{P}_+ \subset L^2(\Omega \times [0, T]; U)$, any $\|\cdot\|_0$ -limit g of a sequence $(g_n)_n \subset \mathcal{E}_0$ will determine a well-defined element in $L^2(\Omega \times [0, T]; U)$.

Moreover, as a consequence of this, we will only need to check the equality of the integrals for integrands g in \mathcal{E}_0 . Since such elements are contained in \mathcal{P}_+ , Dalang's integral of g with respect to the martingale measure M turns out to be a Walsh integral, so that we can conclude by using the first part of the proof. \Box

Remark 2.7. According to Proposition 2.6, when one integrates an element of \mathcal{P}_+ , it is possible to use either the Walsh integral or the integral with respect to a cylindrical Wiener process. However, the Walsh integral enjoys additional properties, in part because it is possible to make use of the dominating measure, which can be very useful in certain estimates. For example, establishing Hölder continuity of the solution to the 1-dimensional stochastic wave equation, in which a Walsh integral appears, is an easy exercise [41, Exercise 3.7], while for the 3-dimensional stochastic wave equation, this is quite involved [12].

2.4. Examples of integrands

In this section, we aim to provide useful examples of random distributions which belong to $L^2(\Omega \times [0, T]; U)$, that is, for which we can define the stochastic integral (2.7) with respect to W.

Recall that an element $\Theta \in S'(\mathbb{R}^d)$ is a non-negative distribution with rapid decrease if Θ is a non-negative measure and if

$$\int_{\mathbb{R}^d} (1+|x|^2)^{k/2} \, \Theta(dx) < +\infty,$$

for all k > 0 (see [39]).

Recall that μ is the spectral measure of *W*. We consider the following hypothesis.

Hypothesis 2.8. Let Γ be a function defined on \mathbb{R}_+ with values in $S'(\mathbb{R}^d)$ such that, for all t > 0, $\Gamma(t)$ is a non-negative distribution with rapid decrease, and

$$\int_0^T dt \, \int_{\mathbb{R}^d} \mu(d\xi) \, |\mathcal{F}\Gamma(t)(\xi)|^2 < \infty. \tag{2.11}$$

In addition, Γ is a non-negative measure of the form $\Gamma(t, dx) dt$ such that, for all T > 0,

$$\sup_{0\leq t\leq T}\Gamma(t,\mathbb{R}^d)<\infty$$

The main examples of integrands are provided by the following proposition (see [29, Proposition 3.3 and Remark 3.4]). In comparison with the analogous result by Dalang [6, Theorem 2], Proposition 2.9 does not require that the stochastic process Z has a spatially homogeneous covariance (see Hypothesis A in [6]).

Proposition 2.9. Assume that Γ satisfies Hypothesis 2.8. Let $Z = \{Z(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ be a predictable process such that

$$\sup_{(t,x)\in[0,T]\times\mathbb{R}^d} E(|Z(t,x)|^p) < \infty,$$
(2.12)

for some $p \ge 2$. Then, the random measure $G = \{G(t, dx) = Z(t, x)\Gamma(t, dx), t \in [0, T]\}$ is a predictable process with values in $L^p(\Omega \times [0, T]; U)$. Moreover,

$$E\left(\|G\|_{U_T}^2\right) = E\left[\int_0^T dt \int_{\mathbb{R}^d} \mu(d\xi) \left|\mathcal{F}(\Gamma(t)Z(t))(\xi)\right|^2\right]$$

and

$$E\left(|G \cdot W|^{p}\right) \leq C \int_{0}^{T} dt \left(\sup_{x \in \mathbb{R}^{d}} E(|Z(t,x)|^{p})\right) \int_{\mathbb{R}^{d}} \mu(d\xi) \left|\mathcal{F}\Gamma(t)(\xi)\right|^{2}.$$

The integral of $G = \{G(t, dx) = Z(t, x)\Gamma(t, dx), t \in [0, T]\}$ with respect to W will be also denoted by

$$G \cdot W = \int_0^T \int_{\mathbb{R}^d} \Gamma(s, y) Z(s, y) W(ds, dy).$$
(2.13)

It is worth pointing out two key steps in the proof of this proposition (see [29]): the first is to check that under Hypothesis 2.8, Γ belongs to $U_T = L^2([0, T]; U)$; the second is to notice that if Γ and Z satisfy, respectively, Hypothesis 2.8 and condition (2.12), then $G(t) = Z(t, \star)\Gamma(t, \star)$ defines a distribution with rapid decrease, almost surely.

Remark 2.10. We note that [5] presents a further extension of Walsh's stochastic integral, with which it becomes possible to integrate certain random elements of the form $Z(t, \star)\Gamma(t, \star)$, where Γ is a tempered distribution which is not necessarily non-negative. This extension is useful for studying the stochastic wave equation in high spatial dimensions.

2.5. The Dalang–Mueller extension of the stochastic integral

We briefly summarize here the function-valued stochastic integral constructed in [9]. This is an extension of Walsh's stochastic integral, where one integrates processes that take values in $L^2(\mathbb{R}^d)$ (or a weighted L^2 -space) and the value of the integral is in the same L^2 -space.

Suppose that $s \mapsto \Gamma(s) \in \mathcal{S}'(\mathbb{R}^d)$ satisfies:

(1) For all $s \ge 0$, $\mathcal{F}\Gamma(s)$ is a function and

$$\int_0^T ds \sup_{\xi \in \mathbb{R}^d} \int_{\mathbb{R}^d} \mu(d\eta) |\mathcal{F}\Gamma(s)(\xi-\eta)|^2 < +\infty.$$

- (2) For all $\phi \in C_0^{\infty}(\mathbb{R}^d)$, sup $_{0 \le s \le T} \Gamma(s) * \phi$ is a bounded function on \mathbb{R}^d . Suppose that $s \mapsto Z(s) \in L^2(\mathbb{R}^d)$ satisfies:
- (3) For $0 \le s \le T$, $Z(s) \in L^2(\mathbb{R}^d)$ a.s., Z(s) is \mathcal{F}_s -measurable, and $s \mapsto Z(s)$ is mean-square continuous from [0, T] into $L^2(\mathbb{R}^d)$.

For such Γ and Z, one sets

$$I_{\Gamma,Z} := \int_0^T ds \int_{\mathbb{R}^d} d\xi \, E\left(|\mathcal{F}Z(s)(\xi)|^2\right) \int_{\mathbb{R}^d} \mu(d\eta) |\mathcal{F}\Gamma(s)(\xi-\eta)|^2 < +\infty.$$
(2.14)

Then the stochastic integral

$$v_{\Gamma,Z} = \int_0^T \int_{\mathbb{R}^d} \Gamma(s, \star -y) Z(s, y) M(ds, dy)$$
(2.15)

is defined as an element of $L^2(\Omega \times \mathbb{R}^d, dP \times dx)$, such that

$$E\left(\left\|v_{\Gamma,Z}\right\|_{L^{2}(\mathbb{R}^{d})}^{2}\right) = I_{\Gamma,Z}.$$
(2.16)

This definition is obtained in three steps.

(a) If, in addition to (1), $\Gamma(s) \in C^{\infty}(\mathbb{R}^d)$, for $0 \le s \le T$, and in addition to (3), $Z(s) \in C_0^{\infty}(\mathbb{R}^d)$ and there is a compact $K \subset \mathbb{R}^d$ such that supp $Z(s) \subset K$, for $0 \le s \le T$, then

$$v_{\Gamma,Z}(x) = \int_0^T \int_{\mathbb{R}^d} \Gamma(s, x - y) Z(s, y) M(ds, dy),$$

where the right-hand side is a Walsh stochastic integral. Equality (2.16) is checked by direct calculation (see [9, Lemma 1]).

(b) If Γ is as in (a) and Z satisfies (3), then one checks that

$$\lim_{m\to\infty}\lim_{n\to\infty}I_{\Gamma,Z-(Z1_{[-m,m]})*\psi_n}=0,$$

where $(\psi_n) \subset C_0^{\infty}(\mathbb{R}^d)$ is a sequence that converges to the Dirac distribution, and one sets

$$v_{\Gamma,Z} = \lim_{m\to\infty} \lim_{n\to\infty} v_{\Gamma,(Z1_{[-m,m]})*\psi_n},$$

where the limits are in $L^2(\Omega \times \mathbb{R}^d, dP \times dx)$.

(c) If Γ satisfies (1) and (2), and Z satisfies (3), then one checks that

$$\lim_{n\to\infty}I_{\Gamma-\Gamma*\psi_n,Z}=0$$

and one sets

$$v_{\Gamma,Z} = \lim_{n \to \infty} v_{\Gamma * \psi_n, Z},$$

where the limit is in $L^2(\Omega \times \mathbb{R}^d, dP \times dx)$: see [9, Theorem 6].

In comparison with the stochastic integral of Section 2.3, we remark that the process *Z* verifies $\sup_{s \in [0,T]} E(||Z(s)||^2_{L^2(\mathbb{R}^d)}) < +\infty$, rather than (2.12), and the resulting integral $v_{\Gamma,Z}$, as a random function of *x*, belongs to $L^2(\Omega \times \mathbb{R}^d)$.

We now relate this stochastic integral to the one defined in Section 2.3.

Proposition 2.11. Assume that Γ and Z satisfy conditions (1), (2) and (3) above. Then:

(i) For almost all $x \in \mathbb{R}^d$, the element $\Gamma(\cdot, x - \star)Z(\cdot, \star)$ belongs to $L^2(\Omega \times [0, T]; U)$. Hence, as in (2.7), we can define the (real-valued) stochastic integral

$$\mathcal{I}_{\Gamma,Z}(T,x) := \int_0^T \int_{\mathbb{R}^d} \Gamma(s, x - y) Z(s, y) W(ds, dy), \quad \text{for } a.a. x \in \mathbb{R}^d.$$

(ii) $\mathcal{I}_{\Gamma,Z}(T, \star) \in L^2(\Omega \times \mathbb{R}^d)$ and $\|\mathcal{I}_{\Gamma,Z}(T, \star)\|_{L^2(\Omega \times \mathbb{R}^d)}^2 = I_{\Gamma,Z}$.

(iii) $\mathcal{I}_{\Gamma,Z}(T, \star) = v_{\Gamma,Z} \text{ in } L^2(\Omega \times \mathbb{R}^d).$

Proof. We will split the proof in three steps, which essentially correspond to the construction of the Dalang–Mueller integral $v_{\Gamma,Z}$.

Step 1. Let us assume first that Γ and Z satisfy the hypotheses in (a) above. Then, as we pointed out there, for all $x \in \mathbb{R}^d$, the stochastic integral $v_{\Gamma,Z}(x)$ can be defined as a Walsh stochastic integral. Hence, by Proposition 2.6(a), the integrand $(s, y) \mapsto \Gamma(s, x - y)Z(s, y)$ defines an element in $L^2(\Omega \times [0, T]; U)$ and, for all $x \in \mathbb{R}^d$, $v_{\Gamma,Z}(x) = \mathcal{I}_{\Gamma,Z}(T, x)$. Condition (ii) in the statement can be deduced from this latter equality and (2.16).

Step 2. Assume now that Γ is as in Step 1 and Z satisfies condition (3). Then, as in (b) above, there exists a sequence of processes $(Z_n)_n$ such that, for all $n \ge 1$, Z_n satisfies the hypotheses in (a) and I_{Γ, Z_n-Z} converges to zero as n tends to infinity. For this sequence,

$$v_{\Gamma,Z} := \lim_{n \to \infty} v_{\Gamma,Z_n} = \lim_{n \to \infty} \mathcal{I}_{\Gamma,Z_n}(T,\star)$$
(2.17)

by Step 1, where the limit is in $L^2(\Omega \times \mathbb{R}^d)$.

We now check property (i) in the statement of the proposition. Observe that, by Proposition 2.9,

$$\int_{\mathbb{R}^d} dx \, \|\Gamma(\cdot, x - \star)[Z_n(\cdot, \star) - Z(\cdot, \star)]\|_{L^2(\Omega \times [0, T]; U)}^2$$
$$= \int_{\mathbb{R}^d} dx \, E\left(\int_0^T ds \, \int_{\mathbb{R}^d} \mu(d\eta) |\mathcal{F}(\Gamma(s, x - \star)[Z_n(s, \star) - Z(s, \star)])(\eta)|^2\right). \tag{2.18}$$

Use the very last lines in the proof of [9, Lemma 1] to see that this is equal to I_{Γ,Z_n-Z} . Since this quantity converges to zero as $n \to \infty$, we deduce that there exists a subsequence $(n_j)_j$ such that, for almost all $x \in \mathbb{R}^d$,

$$\lim_{j\to\infty} \|\Gamma(\cdot, x-\star)Z_{n_j}(\cdot, \star) - \Gamma(\cdot, x-\star)Z(\cdot, \star)\|_{L^2(\Omega\times[0, T]; U)} = 0.$$

This implies that, for almost all $x \in \mathbb{R}^d$, the element $(s, y) \mapsto \Gamma(s, x - y)Z(s, y)$ belongs to $L^2(\Omega \times [0, T]; U)$, and we can define the (real-valued) stochastic integral

$$\mathcal{I}_{\Gamma,Z}(T,x) := \int_0^T \int_{\mathbb{R}^d} \Gamma(s, x - y) Z(s, y) W(ds, dy),$$
(2.19)

and

$$\mathcal{I}_{\Gamma,Z}(T,x) = \lim_{j \to \infty} \mathcal{I}_{\Gamma,Z_{n_j}}(T,x) \quad \text{in} \quad L^2(\Omega)$$

Notice that

$$\|\mathcal{I}_{\Gamma,Z_n}(T,\star) - \mathcal{I}_{\Gamma,Z}(T,\star)\|_{L^2(\Omega \times \mathbb{R}^d)}^2 = \|\mathcal{I}_{\Gamma,Z_n-Z}(T,\star)\|_{L^2(\Omega \times \mathbb{R}^d)}^2.$$
(2.20)

By the isometry property (2.8), this is equal to (2.18), and therefore to I_{Γ, Z_n-Z} , which tends to 0 as $n \to \infty$. Therefore, using Step 1, we see that

$$\|\mathcal{I}_{\Gamma,Z}(T,\star)\|_{L^{2}(\Omega\times\mathbb{R}^{d})}^{2} = \lim_{n\to\infty}\|\mathcal{I}_{\Gamma,Z_{n}}(T,\star)\|_{L^{2}(\Omega\times\mathbb{R}^{d})}^{2} = \lim_{n\to\infty}I_{\Gamma,Z_{n}}=I_{\Gamma,Z},$$

which proves (ii). The arguments following (2.20) and (2.18) prove (iii).

Step 3. In this final part, we assume that Γ and Z satisfy conditions (1), (2) and (3). Then, it is a consequence of step (c) above that there exists $(\Gamma_n)_n$ such that, for all $n \ge 1$, Γ_n verifies the assumptions of the previous step and

$$\lim_{n\to\infty}I_{\Gamma_n-\Gamma,Z}=0.$$

In order to prove parts (i), (ii) and (iii) for this case, one can follow exactly the same lines as we have done in Step 2. We omit the details. \Box

As we will explain in Section 4.6, for the particular case of the stochastic wave equation, it is useful to consider stochastic integrals of the form $v_{\Gamma,Z}$ which take values in some weighted L^2 -space. We now describe this situation.

Fix k > d and let $\theta : \mathbb{R}^d \to \mathbb{R}$ be a smooth function for which there are constants 0 < c < C such that

$$c(1 \wedge |x|^{-k}) \le \theta(x) \le C(1 \wedge |x|^{-k}).$$

The weighted L^2 -space L^2_{θ} is the set of measurable $g : \mathbb{R}^d \to \mathbb{R}$ such that $\|g\|_{\theta} < +\infty$, where

$$\|g\|_{\theta}^2 = \int_{\mathbb{R}^d} |g(x)|^2 \,\theta(x) \, dx.$$

Consider a function $s \mapsto \Gamma(s) \in \mathcal{S}'(\mathbb{R}^d)$ that satisfies (1), (2) above, and, in addition,

(4) There is R > 0 such that for $s \in [0, T]$, supp $\Gamma(s) \subset B(0, R)$.

For a stochastic process *Z*, we consider the following hypothesis:

(5) For $0 \le s \le T$, $Z(s) \in L^2_{\theta}$ a.s., Z(s) is \mathcal{F}_s -measurable, and $s \mapsto Z(s)$ is mean-square continuous from [0, T] into L^2_{θ} .

Then the stochastic integral

$$v_{\Gamma,Z}^{\theta} = \int_0^T \int_{\mathbb{R}^d} \Gamma(s, \star -y) Z(s, y) M(ds, dy)$$
(2.21)

is defined as an element of $L^2(\Omega \times \mathbb{R}^d, dP \times \theta(x)dx)$, such that

$$E(\|v_{\Gamma,Z}^{\theta}\|_{L^2_{\theta}}^2) \leq I_{\Gamma,Z}^{\theta},$$

where

$$I_{\Gamma,Z}^{\theta} := \int_0^T ds \ E(\|Z(s,\star)\|_{L^2_{\theta}}^2) \sup_{\xi \in \mathbb{R}^d} \int_{\mathbb{R}^d} \mu(d\eta) |\mathcal{F}\Gamma(s)(\xi-\eta)|^2.$$

This definition is obtained by showing that $Z_n(s, \star) := Z(s, \star) \mathbb{1}_{[-n,n]}(\star)$ also satisfies (5) as well as (3). Therefore, $v_{\Gamma,Z_n}^{\theta} = v_{\Gamma,Z_n}$ is defined as an element of $L^2(\Omega \times \mathbb{R}^d, dP \times dx)$, and one checks that this element also belongs to $L^2(\Omega \times \mathbb{R}^d, dP \times \theta(x)dx)$, and

$$\lim_{n\to\infty}I^{\theta}_{\Gamma,Z-Z_n}=0,$$

provided that Γ satisfies (1), (2) and (4). Then one sets

$$v_{\Gamma,Z}^{\theta} = \lim_{n \to \infty} v_{\Gamma,Z_n},$$

where the limit is in $L^2(\Omega \times \mathbb{R}^d, dP \times \theta(x)dx)$: see [9, Theorem 12].

2.6. Hilbert-space-valued integrals and tensor products

In this section, we return to the general setting of Section 2.1 and we explain how the real-valued stochastic integral defined there can be naturally extended to a Hilbert-space-valued integral. In Section 3.6, we will show that this extended stochastic integral is equivalent to the stochastic integral of Da Prato and Zabczyk [14], that we will present in Section 3.3.

As far as we know, the stochastic integral that we present here does not appear explicitly in the literature. Nevertheless, in the particular case where the cylindrical Wiener process is given by the spatially homogeneous noise of Section 2.2, a definition of such Hilbert-space-valued integrals has been given in [35], in [38, Chapter 6], and in [29, Section 3] (for a particular form of the integrands, as in Proposition 2.12 below). For related papers where this type of integral has also been used, we refer the reader for instance to [24,26,36].

Let *V* and *H* be Hilbert spaces with inner product $\langle \cdot, \cdot \rangle_V$ and $\langle \cdot, \cdot \rangle_H$, respectively, and $B = \{B_t(h), t \ge 0, h \in V\}$ be a cylindrical Wiener process on *V* with covariance *Q* (see Definition 2.1). Recall that V_Q denotes the Hilbert space *V* endowed with the inner product $\langle h, g \rangle_{V_Q} = \langle Qh, g \rangle_V$.

Let $V_Q \otimes H$ be the Hilbert space tensor product of V_Q and H. We recall that if $(v_j)_j$ and $(f_k)_k$ denote complete orthonormal bases of V_Q and H, respectively, then $(v_j \otimes f_k)_{j,k}$ defines a complete orthonormal basis of $V_Q \otimes H$ and any element $X \in V_Q \otimes H$ can be represented in the following forms (see e.g. [42, Section 3.4]):

$$X = \sum_{j,k=1}^{\infty} X^{j,k} (v_j \otimes f_k) = \sum_{k=1}^{\infty} \left(\sum_{j=1}^{\infty} X^{j,k} v_j \right) \otimes f_k,$$

where $X^{j,k} \in \mathbb{R}$ and

$$\sum_{j,k=1}^{\infty} (X^{j,k})^2 < +\infty,$$
(2.22)

so that $||X||^2_{V_Q \otimes H} = \sum_{j,k=1}^{\infty} (X^{j,k})^2$. This representation shows that the tensor product $V_Q \otimes H$ is isomorphic to the set of "matrices" $(X^{j,k})$ satisfying (2.22).

We will define an *H*-valued stochastic integral $g \cdot B$ of any predictable process $g \in L^2(\Omega \times [0, T]; V_Q \otimes H)$. More precisely, note first that if g is such a process, then for all $s \in [0, T]$, $g_s = \sum_{i,k=1}^{\infty} g_s^{i,k} (v_j \otimes f_k)$, and

$$E\left(\int_{0}^{T} \|g_{s}\|_{V_{Q}\otimes H}^{2} ds\right) = E\left(\int_{0}^{T} \sum_{j,k=1}^{\infty} \left(g_{s}^{j,k}\right)^{2} ds\right) < +\infty.$$
(2.23)

For any $k \in \mathbb{N}$, let g^k be the stochastic process

$$g_s^k := \sum_{j=1}^{\infty} g_s^{j,k} v_j, \qquad s \in [0,T].$$

Then g^k defines a predictable element in $L^2(\Omega \times [0, T]; V_Q)$. Indeed, for all *s*, we have that $g_s^k \in V_Q$ a.s., and by (2.23),

$$E\left(\int_0^T \left\|g_s^k\right\|_{V_Q}^2 ds\right) = E\left(\int_0^T \sum_{j=1}^\infty \left(g_s^{j,k}\right)^2 ds\right) \le E\left(\int_0^T \sum_{j,k=1}^\infty \left(g_s^{j,k}\right)^2 ds\right) < +\infty.$$

As in (2.1), this implies that the real-valued stochastic integral $g^k \cdot B$ is well-defined and satisfies

$$E\left((g^k\cdot B)^2\right)=E\left(\int_0^T \left\|g_s^k\right\|_{V_Q}^2\,ds\right).$$

We now define the *H*-valued stochastic integral of g with respect to *B* as follows:

$$g \cdot B := \sum_{k=1}^{\infty} (g^k \cdot B) f_k.$$
(2.24)

We also use the notation

$$g\cdot B=\int_0^T g_s\,dB_s.$$

One easily verifies that the above series converges in $L^2(\Omega; H)$ and the isometry property in this case reads

$$E\left(\left\|g\cdot B\right\|_{H}^{2}\right)=E\left(\int_{0}^{T}\left\|g_{s}\right\|_{V_{Q}\otimes H}^{2}ds\right).$$

We now provide some examples of processes that can be integrated in the sense just defined. For this, we consider the spatially homogeneous Gaussian noise defined in Section 2.2, so that we use the Hilbert space *U* introduced in Section 2.2 and the standard cylindrical Wiener process $W = \{W_t(\varphi), t \ge 0, \varphi \in U\}$ defined in Proposition 2.5. We denote by $(e_j)_j$ a complete orthonormal basis of *U*. We have the following Hilbert-space-valued counterpart of Proposition 2.9. Its proof follows the same lines as the analogous result for real-valued integrals (see [29, Proposition 3.3 and (3.13)]).

Proposition 2.12. Assume that Γ satisfies Hypothesis 2.8. Let $K = \{K(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ be a predictable process with values in H such that

 $\sup_{(t,x)\in[0,T]\times\mathbb{R}^d}E(\|K(t,x)\|_H^p)<\infty,$

for some $p \ge 2$. Then the element $G = \{G(t, dx) = K(t, x)\Gamma(t, dx), t \in [0, T]\}$ is a predictable process with values in $L^p(\Omega \times [0, T]; U \otimes H)$. Moreover,

$$E\left(\int_0^T \|G_t\|_{U\otimes H}^2 dt\right) = \sum_{k=1}^\infty E\left[\int_0^T dt \int_{\mathbb{R}^d} \mu(d\xi) |\mathcal{F}(G^k(t))(\xi)|^2\right],$$

where $G^k \in L^2(\Omega \times [0, T]; U)$ is the predictable process defined by $G^k := \{G^k(t, dx) = K^k(t, x) \Gamma(t, dx), (t, x) \in [0, T] \times \mathbb{R}^d\}$, with $K^k(t, x) = \langle K(t, x), f_k \rangle_H$, and

$$E\left(\|G\cdot B\|_{H}^{p}\right) \leq C\int_{0}^{T} dt \left(\sup_{x\in\mathbb{R}^{d}} E(\|K(t,x)\|_{H}^{p})\right) \int_{\mathbb{R}^{d}} \mu(d\xi) |\mathcal{F}\Gamma(t)(\xi)|^{2}.$$

3. Infinite-dimensional integration theory

In this section, we sketch in Sections 3.1–3.3 the construction of the infinite-dimensional stochastic integral in the setup of Da Prato and Zabczyk in [14]. For this, we will define the general concept of Hilbert-space-valued Q-Wiener process and study its relationship with the cylindrical Wiener process considered in Section 2.1. Then we will show in Sections 3.4 and 3.5 that the stochastic integral constructed in Section 2.1 can be inserted into this more abstract setting. In particular, we will treat specifically the case of the standard cylindrical Wiener process given by the spatially homogeneous noise described in Section 2.2. In Section 3.6, we establish the equivalence between the Hilbert-space-valued integral of Section 2.6 and the stochastic integral of Sections 3.3 and 3.5.

We begin by recalling some facts concerning nuclear and Hilbert–Schmidt operators on Hilbert spaces.

3.1. Nuclear and Hilbert–Schmidt operators

Let *E*, *G* be Banach spaces and let L(E, G) be the vector space of all linear bounded operators from *E* into *G*. We denote by E^* and G^* the dual spaces of *E* and *G*, respectively.

An element $T \in L(E, G)$ is said to be a *nuclear* operator if there exist two sequences $(a_j)_j \subset G$ and $(\varphi_j)_j \subset E^*$ such that

$$T(x) = \sum_{j=1}^{\infty} a_j \varphi_j(x), \quad \text{for all } x \in E,$$

and

$$\sum_{j=1}^{\infty} \left\|a_j\right\|_G \left\|\varphi_j\right\|_{E^*} < +\infty.$$

The space of all nuclear operators from E into G is denoted by $L_1(E, G)$. When endowed with the norm

$$||T||_{1} = \inf \left\{ \sum_{j=1}^{\infty} ||a_{j}||_{G} ||\varphi_{j}||_{E^{*}} : T(x) = \sum_{j=1}^{\infty} a_{j}\varphi(x), x \in E \right\},\$$

it is a Banach space.

Let *H* be a separable Hilbert space and let $(e_j)_j$ be a complete orthonormal basis in *H*. For $T \in L_1(H, H)$, the *trace* of *T* is

$$\operatorname{Tr} T = \sum_{j=1}^{\infty} \langle T(e_j), e_j \rangle_H.$$
(3.1)

One proves that if $T \in L_1(H) := L_1(H, H)$, then Tr *T* is a well-defined real number and its value does not depend on the choice of the orthonormal basis (see, for instance, [14, Proposition C.1]). Further, according to [14, Proposition C.3], a non-negative definite operator $T \in L(H)$ is nuclear if and only if, for an orthonormal basis (e_i)_i on *H*,

$$\sum_{j=1}^{\infty} \left\langle T(e_j), e_j \right\rangle_H < +\infty.$$

Moreover, in this case, $\operatorname{Tr} T = ||T||_1$.

Let *V* and *H* be two separable Hilbert spaces and $(e_k)_k$ a complete orthonormal basis of *V*. A bounded linear operator $T: V \to H$ is said to be *Hilbert–Schmidt* if

$$\sum_{k=1}^{\infty} \|T(e_k)\|_H^2 < +\infty.$$

It turns out that the above property is independent of the choice of the basis in V. The set of Hilbert–Schmidt operators from V into H is denoted by $L_2(V, H)$. The norm in this space is defined by

$$\|T\|_{2} = \left(\sum_{k=1}^{\infty} \|T(e_{k})\|_{H}^{2}\right)^{1/2},$$
(3.2)

and defines a Hilbert space with inner product

$$\langle S, T \rangle_2 = \sum_{k=1}^{\infty} \langle S(e_k), T(e_k) \rangle_H.$$
(3.3)

Finally, let us point out that (3.1) and (3.2) imply that if $T \in L_2(V, H)$, then $TT^* \in L_1(H)$, where T^* is the adjoint operator of T, and

$$\|T\|_{2}^{2} = \operatorname{Tr}(TT^{*}). \tag{3.4}$$

We conclude this section by recalling the definition and some properties of the pseudo-inverse of bounded linear operators (see, for instance, [33, Appendix C]).

Let $T \in L(V, H)$ and Ker $T := \{x \in V : T(x) = 0\}$. The *pseudo-inverse* of the operator T is defined by

$$T^{-1} := \left(T_{|_{(\operatorname{Ker} T)^{\perp}}}\right)^{-1} : T(V) \to (\operatorname{Ker} T)^{\perp}.$$

Notice that T is one-to-one on $(\text{Ker }T)^{\perp}$ (the orthogonal complement of Ker T) and T^{-1} is linear and bijective.

If $T \in L(V)$ is a bounded linear operator defined on V and T^{-1} denotes the pseudo-inverse of T, then (see [33, Proposition C.0.3]):

1. $(T(V), \langle \cdot, \cdot \rangle_{T(V)})$ defines a Hilbert space, where

$$\langle x, y \rangle_{T(V)} := \langle T^{-1}(x), T^{-1}(y) \rangle_V, \qquad x, y \in T(V).$$

2. Let $(e_k)_k$ be an orthonormal basis of $(\text{Ker } T)^{\perp}$. Then $(T(e_k))_k$ is an orthonormal basis of $(T(V), \langle \cdot, \cdot \rangle_{T(V)})$.

Finally, according to [33, Corollary C.0.6], if $T \in L(V, H)$ and we set $Q := TT^* \in L(H)$, then we have Im $Q^{1/2} = \text{Im } T$ and

$$\|Q^{-1/2}(x)\|_{H} = \|T^{-1}(x)\|_{V}, \quad x \in \operatorname{Im} T,$$

where $Q^{-1/2}$ is the pseudo-inverse of $Q^{1/2}$.

3.2. Hilbert-space-valued Wiener processes

The stochastic integral presented in Da Prato and Zabczyk [14] is defined with respect to a class of Hilbert-space-valued processes, namely Q-Wiener processes, which we now introduce.

We consider a separable Hilbert space V and a linear, symmetric (self-adjoint) non-negative definite and bounded operator Q on V such that Tr $Q < +\infty$.

Definition 3.1. A *V*-valued stochastic process { W_t , $t \ge 0$ } is called a *Q*-Wiener process if (1) $W_0 = 0$, (2) W has continuous trajectories, (3) W has independent increments, and (4) the law of $W_t - W_s$ is Gaussian with mean zero and covariance operator (t - s)*Q*, for all $0 \le s \le t$.

We recall that according to [14, Section 2.3.2], condition (4) above means that for any $h \in V$ and $0 \le s \le t$, the real-valued random variable $\langle W_t - W_s, h \rangle_V$ is Gaussian, with mean zero and variance $(t - s) \langle Qh, h \rangle_V$. In particular, using (3.1), we see that $E(||W_t||_V^2) = t$ Tr Q, which is one reason why the assumption Tr $Q < \infty$ is essential.

Let $(e_j)_j$ be an orthonormal basis of *V* that consists of eigenvectors of *Q* with corresponding eigenvalues $\lambda_j, j \in \mathbb{N}^*$. Let $(\beta_j)_j$ be a sequence of independent real-valued standard Brownian motions on a probability space (Ω, \mathcal{F}, P) . Then the *V*-valued process

$$\mathcal{W}_t = \sum_{j=1}^{\infty} \sqrt{\lambda_j} \,\beta_j(t) e_j \tag{3.5}$$

(where the series converges in $L^2(\Omega; \mathcal{C}([0, T]; V)))$, defines a Q-Wiener process on V (see (2.1.2) in [33]). We note that $\sqrt{\lambda_j} e_j = Q^{1/2}(e_j)$. In the special case where V is finite-dimensional, say dim V = n, then Q can be identified with an $n \times n$ -matrix which is the variance-covariance matrix of $\{W_t\}$, and $\{W_t\}$ has the same law as $\{Q^{1/2}W_t^0\}$, where $\{W_t^0\}$ is a standard Brownian motion with values in \mathbb{R}^n .

If { W_t , $t \ge 0$ } is a Q-Wiener process on V, there is a natural way to associate to it a cylindrical Wiener process in the sense of Definition 2.1. Namely, for any $h \in V$ and $t \ge 0$, we set $W_t(h) := \langle W_t, h \rangle_V$. Using polarization, one checks that { $W_t(h)$, $t \ge 0$, $h \in V$ } is a cylindrical Wiener process on V with covariance operator Q. Note that in this case, $W_t(e_j) = \sqrt{\lambda_j} \beta_j(t)$, so the Brownian motions β_j in (3.5) are given by $\beta_j(t) = W_t(v_j)$, where

$$v_j = \lambda_j^{-1/2} e_j = Q^{-1/2}(e_j), \quad \text{for } j \ge 1 \quad \text{with} \quad \lambda_j \ne 0.$$
 (3.6)

In particular, $(v_j)_j$ is a complete orthonormal basis of the space V_Q of Section 2.1.

However, it is not true in general that any cylindrical Wiener process is associated to a Q-Wiener process on a Hilbert space. Indeed, we have the following result (see [25, p. 177]).

Theorem 3.2. Let V be a separable Hilbert space and W a cylindrical Wiener process on V with covariance Q. Then, the following three conditions are equivalent:

- 1. W is associated to a V-valued Q-Wiener process W, in the sense that $\langle W_t, h \rangle_V = W_t(h)$, for all $h \in V$.
- 2. For any $t \ge 0$, $h \mapsto W_t(h)$ defines a Hilbert–Schmidt operator from V into $L^2(\Omega, \mathcal{F}, P)$.

3. Tr
$$Q < +\infty$$
.

If any one of the above conditions holds, then the norm of the Hilbert–Schmidt operator $h \mapsto W_t(h)$, as an element of $L_2(V, L^2(\Omega, \mathcal{F}, P))$, is given by

$$||W_t||_2 = E(||W_t||_V^2) = t \operatorname{Tr} Q.$$

As a consequence of the above result, if dim $V = +\infty$ and if W is a standard cylindrical Wiener process on V, that is $Q = Id_V$, then there is no Q-Wiener process W associated to W. However, as we will explain in Section 3.5, it will be possible to find a Hilbert-space-valued Wiener process with values in a larger Hilbert space V_1 which will correspond to W in a certain sense.

3.3. H-valued stochastic integrals

We now sketch the construction of the infinite-dimensional stochastic integral of [14]. Let *V* and *H* be two separable Hilbert spaces and let { W_t , $t \ge 0$ } be a *Q*-Wiener process defined on *V*. We denote by (\mathcal{F}_t)_t the (completed) filtration generated by \mathcal{W} . In [14], the objective is to construct the *H*-valued stochastic integral

$$\int_0^t \Phi_s \, d\mathcal{W}_s, \qquad t \in [0, T],$$

where Φ is a process with values in the space of linear but not necessarily bounded operators from V into H.

Consider the subspace $V_0 := Q^{1/2}(V)$ of V which, endowed with the inner product

$$\langle h,g\rangle_0 := \langle Q^{-1/2}h,Q^{-1/2}g\rangle_V,$$

is a Hilbert space. Here $Q^{-1/2}$ denotes the pseudo-inverse of the operator $Q^{1/2}$ (see Section 3.1). Let us also set

$$L_2^0 := L_2(V_0, H),$$

which is the Hilbert space of all Hilbert–Schmidt operators from V_0 into H, equipped, as in (3.3), with the inner product

$$\langle \Phi, \Psi \rangle_{L_2^0} = \sum_{j=1}^{\infty} \langle \Phi \tilde{e}_j, \Psi \tilde{e}_j \rangle_H, \qquad \Phi, \Psi \in L_2^0,$$
(3.7)

where $(\tilde{e}_i)_i$ is any complete orthonormal basis of V_0 . In particular, using the fact that we can take

$$\tilde{e}_j = \sqrt{\lambda_j} e_j = Q^{1/2}(e_j), \qquad j \ge 1, \ \lambda_j > 0, \tag{3.8}$$

where the $(e_j)_j$ are as in (3.5) (see condition 2. in the final part of Section 3.1) and applying (3.4), the norm of $\Psi \in L^0_2$ can be expressed as

$$\|\Psi\|_{L_{2}^{0}}^{2} = \|\Psi \circ Q^{1/2}\|_{L_{2}(V,H)}^{2} = \operatorname{Tr}(\Psi Q \Psi^{*}).$$

We note that in the case where dim $V = n < +\infty$ and dim $H = m < +\infty$, then it is natural to identify $\Psi \in L_2^0$ with an $m \times n$ -matrix and Q with an $n \times n$ -matrix. The norm of Ψ corresponds to a classical matrix norm of $\Psi Q^{1/2}$ (whose square is the sum of squares of entries of $\Psi Q^{1/2}$).

Let $\Phi = \{\Phi_t, t \in [0, T]\}$ be a measurable L_2^0 -valued process. We define the norm of Φ by

$$\|\Phi\|_{T} := \left[E\left(\int_{0}^{T} \|\Phi_{s}\|_{L^{0}_{2}}^{2} ds\right) \right]^{1/2}.$$

The aim of [14, Chapter 4], is to define the stochastic integral with respect to W of any L_2^0 -valued predictable process Φ such that $\|\Phi\|_T < \infty$. More precisely, Da Prato and Zabczyk first consider *simple* processes, which are of the form $\Phi_t = \Phi_0 \mathbf{1}_{(a,b]}(t)$, where Φ_0 is any \mathcal{F}_a -measurable L(V, H)-valued random variable and $0 \le a < b \le T$. For such processes, the stochastic integral takes values in H and is defined by the formula

$$\int_0^t \Phi_s \, d\mathcal{W}_s := \Phi_0(\mathcal{W}_{b\wedge t} - \mathcal{W}_{a\wedge t}), \qquad t \in [0, T].$$
(3.9)

The map $\Phi \mapsto \int_0^{\cdot} \Phi_s dW_s$ is an isometry between the set of simple processes and the space \mathcal{M}_H of square-integrable *H*-valued (\mathcal{F}_t)-martingales $X = \{X_t, t \in [0, T]\}$ endowed with the norm ||X|| =

 $[E(||X_T||_H^2)]^{1/2}$. Indeed, as it is proved in [14] (see also [33, Proposition 2.3.5]), the isometry property for simple processes reads

$$E\left(\left\|\int_{0}^{T}\Phi_{t}\,d\mathcal{W}_{t}\right\|_{H}^{2}\right)=\|\Phi\|_{T}^{2}.$$
(3.10)

Remark 3.3. The appearance of $\|\cdot\|_T$ can be understood by considering the case where $\Phi(t) = \Phi_0 \mathbb{1}_{(a,b]}(t)$, where $\Phi_0 \in L(V, H)$ is deterministic and $0 \le a < b \le T$. Indeed, in this case, using (3.9) and the representation (3.5),

$$E\left(\left\|\int_{0}^{T} \Phi_{t} d\mathcal{W}_{t}\right\|_{H}^{2}\right) = E\left(\left\|\sum_{j} \sqrt{\lambda_{j}} \left(\beta_{j}(b) - \beta_{j}(a)\right) \Phi_{0}(e_{j})\right\|_{H}^{2}\right),$$

and the right-hand side is equal to

$$\begin{split} \sum_{j} \lambda_{j} \left(b - a \right) \left\| \Phi_{0}(e_{j}) \right\|_{H}^{2} &= \left(b - a \right) \sum_{j} \left\| \Phi_{0}(Q^{1/2}e_{j}) \right\|_{H}^{2} \\ &= \left(b - a \right) \left\| \Phi_{0} \circ Q^{1/2} \right\|_{L_{2}(V,H)}^{2} = E\left(\int_{0}^{T} \left\| \Phi_{s} \right\|_{L_{2}^{0}}^{2} ds \right). \end{split}$$

Once the isometry property (3.10) is established, a completion argument is used to extend the above definition to all L_2^0 -valued predictable processes Φ satisfying $\|\Phi\|_T < \infty$. The integral of Φ is denoted by

$$\Phi \cdot \mathcal{W} = \int_0^T \Phi_t \, d\mathcal{W}_t$$

and the isometry property (3.10) is preserved for such processes:

$$E(\|\boldsymbol{\Phi}\cdot\boldsymbol{\mathcal{W}}\|_{H}^{2}) = \|\boldsymbol{\Phi}\|_{T}^{2}.$$

The details of this construction can be found in [14, Chapter 4].

Let us conclude this section by providing a representation of the stochastic integral $\Phi \cdot W$ in terms of ordinary Itô integrals of real-valued processes. Indeed, observe first that the expansion (3.5) can be rewritten in the form

$$\mathcal{W}_t = \sum_{j=1}^{\infty} \beta_j(t) \tilde{e}_j, \tag{3.11}$$

where $(\tilde{e}_j)_j$ is defined in (3.8).

Proposition 3.4. Let $(f_k)_k$ be a complete orthonormal basis in the Hilbert space H. Assume that $\Phi = \{\Phi_t, t \in [0, T]\}$ is any L_2^0 -valued predictable process such that $\|\Phi\|_T < \infty$. Then

$$\int_0^T \Phi_t \, d\mathcal{W}_t = \sum_{k=1}^\infty \left(\sum_{j=1}^\infty \int_0^T \left\langle \Phi_t(\tilde{e}_j), f_k \right\rangle_H d\beta_j(t) \right) f_k.$$
(3.12)

We note for future reference that this proposition remains valid even for cylindrical Wiener processes: see Section 3.5 below, and, in particular, Remark 3.9.

Proof of Proposition 3.4. First of all, we will prove that, under the standing hypotheses, the right-hand side of (3.12) is a well-defined element in $L^2(\Omega; H)$. For this, we will check that

$$E\left[\sum_{k=1}^{\infty}\left(\sum_{j=1}^{\infty}\int_{0}^{T}\left\langle \Phi_{t}(\tilde{e}_{j}),f_{k}\right\rangle_{H}d\beta_{j}(t)\right)^{2}\right]=\|\Phi\|_{T}^{2}$$

where the right-hand side is finite, by assumption.

Since $(\beta_i)_i$ is a family of independent standard Brownian motions,

$$E\left[\sum_{k=1}^{\infty}\left(\sum_{j=1}^{\infty}\int_{0}^{T}\left\langle\Phi_{t}(\tilde{e}_{j}),f_{k}\right\rangle_{H}d\beta_{j}(t)\right)^{2}\right]=\sum_{k,j=1}^{\infty}E\left[\left(\int_{0}^{T}\left\langle\Phi_{t}(\tilde{e}_{j}),f_{k}\right\rangle_{H}d\beta_{j}(t)\right)^{2}\right]$$

and the right-hand side is equal to

$$\sum_{k,j=1}^{\infty} \int_{0}^{T} E\left[\left\langle \Phi_{t}(\tilde{e}_{j}), f_{k} \right\rangle_{H}^{2} \right] dt = E\left[\int_{0}^{T} \sum_{j=1}^{\infty} \|\Phi_{t}(\tilde{e}_{j})\|_{H}^{2} dt \right] = E\left[\int_{0}^{T} \|\Phi_{t}\|_{L_{2}^{0}}^{2} dt \right],$$

and the last term is equal to $\|\Phi\|_T^2$. Hence, the series on the right-hand side of (3.12) defines an element in $L^2(\Omega; H)$ and its norm is given by $\|\Phi\|_T$. Therefore, by the isometry property of the stochastic integral (see (3.10)), in order to prove equality (3.12), we only need to check this equality for simple processes. Namely, assume that Φ is of the form $\Phi_t = \Phi_0 \mathbb{1}_{(a,b]}(t)$, where Φ_0 is a \mathcal{F}_a -measurable L(V, H)-valued random variable and $0 \le a < b \le T$. Then, by (3.9),

$$\int_0^T \Phi_t \, d\mathcal{W}_t = \Phi_0(\mathcal{W}_b - \mathcal{W}_a).$$

On the other hand,

$$\sum_{k=1}^{\infty} \left(\sum_{j=1}^{\infty} \int_0^T \left\langle \Phi_t(\tilde{e}_j), f_k \right\rangle_H d\beta_j(t) \right) f_k = \sum_{k,j=1}^{\infty} \left\langle \Phi_0(\tilde{e}_j), f_k \right\rangle_H (\beta_j(b) - \beta_j(a)) f_k$$

and the right-hand side is equal to

$$\sum_{j=1}^{\infty} (\beta_j(b) - \beta_j(a)) \, \Phi_0(\tilde{e}_j) = \Phi_0\left(\sum_{j=1}^{\infty} (\beta_j(b) - \beta_j(a))\tilde{e}_j\right) = \Phi_0(\mathcal{W}_b - \mathcal{W}_a)$$

where the last equality follows from (3.11). The proof is complete. \Box

3.4. The case where $H = \mathbb{R}$

We consider a cylindrical Wiener process W on some separable Hilbert space V with covariance Q, such that Tr $Q < +\infty$. By Theorem 3.2, W is associated to a V-valued Q-Wiener process W. We shall check that the stochastic integral with respect to W, constructed in Section 2.1, is equal to an integral with respect to W, constructed in Section 3.3, when the Hilbert space H in which the integral takes its values is $H = \mathbb{R}$.

In Section 2.1, we defined the Hilbert space V_Q and the stochastic integral

$$g\cdot W=\int_0^T g_s\,dW_s,$$

for any predictable stochastic process $g \in L^2(\Omega \times [0, T]; V_Q)$, with the isometry property

$$E\left((g\cdot W)^2\right) = E\left(\int_0^T \|g_s\|_{V_Q}^2 ds\right)^2$$

For any $s \in [0, T]$ and $g \in L^2(\Omega \times [0, T]; V_Q)$, we define an operator $\Phi_s^g : V \to \mathbb{R}$ by

$$\Phi_s^g(\eta) := \langle g_s, \eta \rangle_V, \qquad \eta \in V.$$
(3.13)

We denote by L_2^0 the set $L_2(V_0, H)$, with $V_0 = Q^{1/2}(V)$ and $H = \mathbb{R}$.

Proposition 3.5. Under the above assumptions, $\Phi^g = \{\Phi^g_s, s \in [0, T]\}$ defines a predictable process with values in $L^0_2 = L_2(V_0, \mathbb{R})$, such that

$$E\left(\int_{0}^{T} \|\boldsymbol{\Phi}_{s}^{g}\|_{L_{2}^{0}}^{2} ds\right) = E\left(\int_{0}^{T} \|\boldsymbol{g}_{s}\|_{V_{Q}}^{2} ds\right).$$
(3.14)

Therefore, the stochastic integral of Φ^g with respect to W can be defined as in Section 3.3 and in fact,

$$\int_0^T \Phi_s^g \, d\mathcal{W}_s = \int_0^T g_s \, d\mathcal{W}_s. \tag{3.15}$$

Proof. We first check (3.14). Let e_j be as in (3.5), \tilde{e}_j be as in (3.8) and v_j be as in (3.6), so that $\tilde{e}_j = Q(v_j)$. By (3.7) with $H = \mathbb{R}$, and by (3.13),

$$\|\Phi_{s}^{g}\|_{2}^{2} = \sum_{j=1}^{\infty} \langle g_{s}, \tilde{e}_{j} \rangle_{V}^{2} = \sum_{j=1}^{\infty} \langle g_{s}, Qv_{j} \rangle_{V}^{2} = \sum_{j=1}^{\infty} \langle g_{s}, v_{j} \rangle_{V_{Q}}^{2} = \|g_{s}\|_{V_{Q}}^{2}.$$

We conclude that (3.14) holds. We note for later reference that this equality $\|\Phi_s^g\|_2 = \|g_s\|_{V_Q}$ remains valid even if Tr $Q = +\infty$.

Since, by hypothesis, the right hand-side of (3.14) is finite, we deduce that Φ^g is a square integrable process with values in L_2^0 and the stochastic integral $\int_0^T \Phi_s^g dW_s$ is well-defined. It remains to prove (3.15). For this, we apply Proposition 3.4 in the following situation: $H = \mathbb{R}$, with

It remains to prove (3.15). For this, we apply Proposition 3.4 in the following situation: $H = \mathbb{R}$, with one basis vector $f_k = 1$, Φ is defined in (3.13), and the sequence of independent standard Brownian motions in (3.12) is given by $\beta_j(t) = W_t(v_j)$. Therefore,

$$\int_0^T \Phi_t^g \, d\mathcal{W}_t = \sum_{j=1}^\infty \int_0^T \Phi_t^g(\tilde{e}_j) \, d\beta_j(t),$$

and the right-hand side is equal to

$$\sum_{j=1}^{\infty} \int_0^T \left\langle g_t, \tilde{e}_j \right\rangle_V dW_t(v_j) = \sum_{j=1}^{\infty} \int_0^T \left\langle g_t, v_j \right\rangle_{V_Q} dW_t(v_j) = \int_0^T g_t dW_t(v_j)$$

This completes the proof. \Box

3.5. The case Tr Q = $+\infty$

In Proposition 2.5, we showed that the covariance operator of the standard cylindrical Wiener process $\{W_t(g), t \ge 0, g \in U\}$ associated with the spatially homogeneous noise that we considered in Section 2.2 is $Q = \text{Id}_U$, which implies that $\text{Tr } Q = +\infty$. Therefore, we cannot make use of Proposition 3.5 since, in this case, there is no *Q*-*Wiener process* associated to *W*. However, there is the related notion of *cylindrical Q*-*Wiener process*, which we now define.

Let $(V, \|\cdot\|_V)$ be a Hilbert space. Let Q be a symmetric non-negative definite and bounded operator on V, possibly such that $\operatorname{Tr} Q = +\infty$. Let $(e_j)_j$ be an orthonormal basis of V that consists of eigenvectors of Q with corresponding eigenvalues $\lambda_j, j \in \mathbb{N}^*$. Define $V_0 = Q^{1/2}(V)$ as in Section 3.3.

It is always possible to find a Hilbert space V_1 and a bounded linear injective operator J: $(V, \|\cdot\|_V) \rightarrow (V_1, \|\cdot\|_{V_1})$ such that the restriction $J_0 = J_{|V_0}$: $(V_0, \|\cdot\|_{V_0}) \rightarrow (V_1, \|\cdot\|_{V_1})$ is Hilbert–Schmidt. Indeed, as explained in [33, Remark 2.5.1], we may choose $V_1 = V$, $\langle \cdot, \cdot \rangle_{V_1} = \langle \cdot, \cdot \rangle_V$, $\alpha_k \in (0, \infty)$ for all $k \ge 1$ such that $\sum_{k=1}^{\infty} \alpha_k^2 < +\infty$, and define $J : V \rightarrow V$ by

$$J(h) := \sum_{k=1}^{\infty} \alpha_k \langle h, e_k \rangle_V e_k, \qquad h \in V,$$
(3.16)

where $(e_k)_k$ is an orthonormal basis of *V*. Then, for $g \in V_0$, $g = \sum_{k=1}^{\infty} \langle g, \tilde{e}_k \rangle_{V_0} \tilde{e}_k$, where $\tilde{e}_k = Q^{1/2}(e_k)$, $k \ge 1$, we have

$$J_0(g) = \sum_{k=1}^{\infty} \alpha_k \langle g, \tilde{e}_k \rangle_{V_0} \sqrt{\lambda_k} \, e_k = \sum_{k=1}^{\infty} \alpha_k \langle g, \tilde{e}_k \rangle_{V_0} \, \tilde{e}_k,$$

and so J_0 : $(V_0, \|\cdot\|_{V_0}) \to (V, \|\cdot\|_V)$ is clearly Hilbert–Schmidt.

As an operator between Hilbert spaces, from V_0 to V_1 , J_0 has an adjoint $J_0^* : V_1 \to V_0$. However, if we consider V_0 and V_1 as Banach spaces, it is more common to consider the adjoint $\tilde{J}_0^* : V_1^* \to V_0^*$.

Proposition 3.6 ([14, Proposition 4.11] and [33, Proposition 2.5.2]).

- 1. Define $Q_1 = J_0 J_0^*$: $V_1 = \text{Im} J_0 \rightarrow V_1$. Q_1 is symmetric (self-adjoint), non-negative definite and Tr $Q_1 < +\infty$.
- 2. Let $\tilde{e}_j = Q^{1/2}(e_j)$, where $(e_j)_j$ is a complete orthonormal basis in *V*, and let $(\beta_j)_j$ be a family of independent real-valued standard Brownian motions. Then

$$\mathcal{W}_t := \sum_{j=1}^{\infty} \beta_j(t) J_0(\tilde{e}_j), \qquad t \ge 0, \tag{3.17}$$

is a Q_1 -Wiener process in V_1 .

3. Let $I : V_0 \rightarrow V_0^*$ be the one-to-one mapping which identifies V_0 with its dual V_0^* , and consider the following diagram:

$$V_1^* \xrightarrow{\tilde{J}_0^*} V_0^{*^{l-1}} V_0 \xrightarrow{J_0} V_1.$$

Then, for all *s*, $t \ge 0$ and $h_1, h_2 \in V_1^*$,

$$E(\langle h_1, \mathcal{W}_s \rangle_1 \langle h_2, \mathcal{W}_t \rangle_1) = (t \wedge s) \langle (I^{-1} \circ \tilde{J}_0^*)(h_1), (I^{-1} \circ \tilde{J}_0^*)(h_2) \rangle_{V_0},$$
(3.18)

where $\langle \cdot, \cdot \rangle_1$ denotes the dual form on $V_1^* \times V_1$.

4. $\operatorname{Im} Q_1^{1/2} = \operatorname{Im} J_0$ and

$$\|h\|_{0} = \|Q_{1}^{-1/2}J_{0}(h)\|_{V_{1}} = \|J_{0}(h)\|_{Q_{1}^{1/2}(V_{1})}, \qquad h \in V_{0}$$

where $Q_1^{-1/2}$ denotes the pseudo-inverse of $Q_1^{1/2}$. Thus, $J_0 : V_0 \rightarrow Q_1^{1/2}(V_1)$ is an isometry.

Remark 3.7. (a) Part 3 in the Proposition's statement is commonly abbreviated in the following formal form (see, for instance, [31, Proposition 1.1]): for all $s, t \ge 0$ and $h_1, h_2 \in V_1^*$,

 $E\left(\langle h_1, \mathcal{W}_s \rangle_1 \langle h_2, \mathcal{W}_t \rangle_1\right) = (t \wedge s) \langle h_1, h_2 \rangle_{V_0}.$

$$E\left(\langle h_1, \mathcal{W}_s \rangle_1 \langle h_2, \mathcal{W}_t \rangle_1\right) = (t \wedge s) \langle Qh_1, h_2 \rangle_V.$$

Proof of Proposition 3.6. Statement 1. follows from (3.4) and the fact that J_0 is Hilbert–Schmidt. Concerning 2., we observe that for $h \in V_1$,

$$E\left(\langle \mathcal{W}_t,h\rangle_{V_1}^2\right)=E\left(\left(\sum_{j=1}^\infty\beta_j(t)\langle J_0(\tilde{e}_j),h\rangle_{V_1}\right)^2\right),$$

and the right-hand side is equal to

$$t\sum_{j=1}^{\infty} \left\langle J_0(\tilde{e}_j), h \right\rangle_{V_1}^2 = t\sum_{j=1}^{\infty} \left\langle \tilde{e}_j, J_0^*(h) \right\rangle_{V_0}^2 = t \left\| J_0^*(h) \right\|_{V_0}^2 = t \left\langle J_0^*(h), J_0^*(h) \right\rangle_{V_0} = t \left\langle J_0 J_0^*(h), h \right\rangle_{V_1}.$$

Let us prove now part 3. For the sake of clarity, we will prove the statement for s = t and $h_1 = h_2$. Hence, let $t \ge 0$ and $h \in V_1^*$. We denote by $\langle \cdot, \cdot \rangle_0$ the dual form on $V_0^* \times V_0$. Then, by (3.17), the relation between J_0 and J_0^* , and the properties of I and the family $(\beta_j)_j$, we obtain

$$E\left(\langle h, \mathcal{W}_t \rangle_1^2\right) = E\left(\left\langle h, \sum_{j=1}^{\infty} \beta_j(t) J_0(\tilde{e}_j) \right\rangle_1^2\right)$$

and the right-hand side is equal to

$$t\sum_{j=1}^{\infty} \left\langle h, J_0(\tilde{e}_j) \right\rangle_1^2 = t\sum_{j=1}^{\infty} \left\langle \tilde{J}_0^*(h), \tilde{e}_j \right\rangle_0^2 = t\sum_{j=1}^{\infty} \left\langle (I^{-1} \circ \tilde{J}_0^*)(h), \tilde{e}_j \right\rangle_{V_0}^2 = t \| (I^{-1} \circ \tilde{J}_0^*)(h) \|_{V_0}^2.$$

For 4., we refer the reader to [33, Proposition 2.5.2]. \Box

Let { W_t , $t \ge 0$ } be as in (3.17). A predictable stochastic process { Φ_t , $t \in [0, T]$ } will be integrable with respect to W if it takes values in $L_2(Q_1^{1/2}(V_1), H)$ and

$$E\left(\int_0^T \|\Phi_t\|_{L_2(Q_1^{1/2}(V_1),H)}^2 dt\right) < +\infty.$$

By part 4 of Proposition 3.6, we have

$$\Phi \in L_2^0 = L_2(V_0, H) \quad \Leftrightarrow \quad \Phi \circ J_0^{-1} \in L_2(Q_1^{1/2}(V_1), H).$$

Definition 3.8. For any square integrable predictable process Φ with values in L_2^0 such that

$$E\left(\int_0^T \|\Phi_t\|_{L^0_2}^2 dt\right) < +\infty,$$

the *H*-valued stochastic integral $\Phi \cdot W$ is defined by

$$\int_0^T \Phi_s \, d\mathcal{W}_s := \int_0^T \Phi_s \circ J_0^{-1} \, d\mathcal{W}_s.$$

Remark 3.9. We note that the class of integrable processes with respect to *W* does not depend on the choice of V_1 , and one checks immediately that even though Tr $Q = +\infty$, formula (3.12) remains valid.

We now relate this notion of stochastic integral with the stochastic integral with respect to the cylindrical Wiener process of Section 2.1. Let $\{W_t, t \in [0, T]\}$ be a cylindrical Wiener process with covariance Q on the Hilbert space V, and let $g \in L^2(\Omega \times [0, T]; V_0)$ be a predictable process, so that $g \cdot W$ is well defined as in Section 2.1. By Proposition 3.6, we can consider the cylindrical Q-Wiener process { W_t , $t \in [0, T]$ } defined by

$$\mathcal{W}_t = \sum_{j=1}^{\infty} \beta_j(t) J_0(\tilde{e}_j)$$
(3.19)

as in formula (3.17) with $\beta_j(t) = W_t(v_j)$, where $v_j = Q^{-1/2}(e_j)$, $\tilde{e}_j = Q^{1/2}(e_j)$ and $(e_j)_j$ denotes a complete orthonormal basis in V consisting of eigenfunctions of Q, so that $(v_j)_j$ is a complete orthonormal basis in V_0 . This process takes values in some Hilbert space V_1 .

For $g \in L^2(\Omega \times [0, T]; V_0)$, we define, as in (3.13), the operator

$$\Phi_s^g(\eta) = \langle g_s, \eta \rangle_V, \qquad \eta \in V,$$

which takes values in $H = \mathbb{R}$. Recall that $V_0 = Q^{1/2}(V)$ and $V_0 = Q^{-1/2}(V)$.

Proposition 3.10. The process $\{\Phi_s^g, s \in [0, T]\}$ defines a predictable process with values in $L_2(V_0, \mathbb{R})$, such that

$$E\left(\int_0^T \|\boldsymbol{\Phi}_s^g\|_2^2\,ds\right) = E\left(\int_0^T \|\boldsymbol{g}_s\|_{V_Q}^2\,ds\right),$$

and

$$\int_0^T \Phi_s^g \, d\mathcal{W}_s = \int_0^T g_s \, dW_s$$

Proof. First, we will prove that $\Phi_s^g \in L_2(V_0, \mathbb{R})$, for $s \in [0, T]$. As in the first part of the proof of Proposition 3.5, $\|\Phi_s^g\|_2 = \|g_s\|_{V_0}$. This gives the equality of expectations in the statement of the proposition, and the right-hand side is finite by assumption.

Concerning the equality of integrals, we note that by definition,

$$\int_0^T \Phi_s \, d\mathcal{W}_s := \int_0^T \Phi_s \circ J_0^{-1} \, d\mathcal{W}_s,$$

where the right-hand side is defined using the finite-trace approach of Section 3.3. We note that by Proposition 3.6, part 4, $(J_0(\tilde{e}_j))_j$ is a complete orthonormal basis of $Q_1^{1/2}(V_1)$. According to Proposition 3.4 with $H = \mathbb{R}$, a single basis element $f_k = 1$ of H, $\beta_j(s) = W_s(v_j)$, and

 \tilde{e}_i there replaced by $J_0(\tilde{e}_i)$, formula (3.12) becomes

$$\int_0^T \Phi_s^g \circ J_0^{-1} d\mathcal{W}_s = \sum_{j=1}^\infty \int_0^T \Phi_s^g \circ J_0^{-1}(J_0(\tilde{e}_j)) d\beta_j(s),$$

and the right-hand side is equal to

$$\sum_{j=1}^{\infty} \int_0^T \Phi_s^g(\tilde{e}_j) \, dW_s(v_j) = \sum_{j=1}^{\infty} \int_0^T \left\langle g_s, \tilde{e}_j \right\rangle_V \, dW_s(v_j) = \sum_{j=1}^{\infty} \int_0^T \left\langle g_s, v_j \right\rangle_{V_Q} \, dW_s(v_j).$$

The last expression is equal to $g \cdot W$. **Remark 3.11.** Proposition 3.10 allows us in particular to associate the spatially homogeneous noise of Section 2.2, viewed as a cylindrical Wiener process with covariance Id_U in Proposition 2.5, with a cylindrical *Q*-Wiener process as in Proposition 3.6, with $Q = Id_U$, on the Hilbert space *U* of Section 2.2, and to relate the associated stochastic integrals.

3.6. Equivalence of Hilbert-space-valued integrals

This section is devoted to proving that the Hilbert-space-valued stochastic integral introduced in Section 2.6 is in fact equivalent to the stochastic integral of Da Prato and Zabczyk described in Sections 3.3 and 3.5.

We consider a cylindrical Wiener process W on a separable Hilbert space V with covariance Q, such that Tr $Q < +\infty$, and let W be the V-valued Q-Wiener process associated to W (see Theorem 3.2). We fix a separable Hilbert space H and a complete orthonormal basis $(f_k)_k$ of H. In Section 2.6, we defined the H-valued stochastic integral $g \cdot W = \int_0^T g_s dW_s$ of any predictable $g \in L^2(\Omega \times [0, T]; V_Q \otimes H)$.

The equivalence that we shall prove is based on the well-known generic fact that the tensor product $V_Q \otimes H$ is canonically isometric and isomorphic to the space $L_2(V_Q^*, H)$ of Hilbert–Schmidt operators defined on the dual space of V_Q and taking values in H (see e.g. [1, Section 12.3]). Notice that the space V_Q^* is non other than $V_0 = Q^{1/2}(V)$. Let $(e_j)_j$ be a complete orthonormal basis of V consisting of eigenvectors of Q. As we have already seen, $v_j = Q^{-1/2}(e_j)$ and $\tilde{e}_j = Q^{1/2}(e_j)$ define complete orthonormal bases of V_Q and V_0 , respectively. If we are given an operator $\Psi \in L_2(V_0, H)$, then its associated element $X^{\Psi} \in V_Q \otimes H$ is given by

$$X^{\Psi} = \sum_{j,k=1}^{\infty} X_{\Psi}^{j,k} (v_j \otimes f_k) \quad \text{with} \quad X_{\Psi}^{j,k} = \left\langle \Psi(\tilde{e}_j), f_k \right\rangle_H.$$

Moreover, one easily checks that $\|\Psi\|_{L_2(V_0, H)} = \|X^{\Psi}\|_{V_0 \otimes H}$.

Proposition 3.12. Let $\Phi = \{\Phi_t, t \in [0, T]\}$ be a predictable process in $L^2(\Omega \times [0, T]; L_2(V_0, H))$, so that we can define the H-valued stochastic integral $\int_0^T \Phi_t dW_t$ in the sense of Da Prato and Zabczyk (see Section 3.3). Let $g^{\Phi} = \{g_t^{\Phi}, t \in [0, T]\}$ be the predictable process in $L^2(\Omega \times [0, T]; V_Q \otimes H)$ which is associated to Φ . That is, for all $t \in [0, T]$ we have

$$g_t^{\Phi} = \sum_{j,k=1}^{\infty} (g_t^{\Phi})^{j,k} (v_j \otimes f_k), \quad \text{with} \quad (g_s^{\Phi})^{j,k} = \left\langle \Phi_s(\tilde{e}_j), f_k \right\rangle_H.$$
(3.20)

Then the H-valued stochastic integral $\int_0^T g_t^{\Phi} dW_t$ of Section 2.6 is well-defined and

$$\int_0^T g_t^{\Phi} \, dW_t = \int_0^T \Phi_t \, d\mathcal{W}_t.$$

Proof. First, note that Proposition 3.4 and the fact that $W_t(h) = \langle W_t, h \rangle_V$, for all $h \in V$, yield

$$\int_0^T \Phi_t \, d\mathcal{W}_t = \sum_{k=1}^\infty \left(\sum_{j=1}^\infty \int_0^T \left\langle \Phi_t(\tilde{e}_j), f_k \right\rangle_H \, dW_t(v_j) \right) f_k.$$

On the other hand, by definition of the integral of g^{ϕ} with respect to W (see (2.24)),

$$\int_0^T g_t^{\Phi} dW_t = \sum_{k=1}^\infty \left[\int_0^T \left(\sum_{j=1}^\infty (g_t^{\Phi})^{j,k} v_j \right) dW_t \right] f_k.$$

Each of the integrals in the above series takes values in \mathbb{R} and is defined by (2.1). Thus

$$\int_0^T g_t^{\varphi} dW_t = \sum_{k=1}^\infty \left(\sum_{j=1}^\infty \int_0^T (g_t^{\varphi})^{j,k} dW_t(v_j) \right) f_k.$$

By (3.20), the proposition is proved. \Box

Remark 3.13. Using Remark 3.9, we see that Proposition 3.12 remains valid in the case where *W* is a cylindrical Wiener process on *V* with covariance *Q* such that $\text{Tr } Q = +\infty$.

4. Spde's driven by a spatially homogeneous noise

This section is devoted to presenting a class of spde's in \mathbb{R}^d driven by a spatially homogeneous noise. In Section 4.1, we present the real-valued approach using the notion of a mild random field solution of the equation. Section 4.2 gives two examples: the stochastic heat equation in any spatial dimension and the stochastic wave dimension in spatial dimensions d = 1, 2, 3. In Section 4.3, we establish an existence and uniqueness result which extends a theorem of [6]. In Section 4.4, we present the infinitedimensional formulation of these spde's. In Section 4.5, we examine the relationship between these two formulations, and conclude that they are equivalent (see Proposition 4.10). In Section 4.6, we examine the relationship with the approach of [9].

We are interested in the following class of non-linear spde's:

$$Lu(t, x) = \sigma(u(t, x))W(t, x) + b(u(t, x)),$$
(4.1)

 $t \ge 0, x \in \mathbb{R}^d$, where *L* denotes a general second order partial differential operator with constant coefficients, with appropriate initial conditions. The coefficients σ and *b* are real-valued functions and $\dot{W}(t, x)$ is the formal notation for the Gaussian random perturbation described at the beginning of Section 2.2.

If *L* is first order in time, such as the heat operator $L = \partial/\partial t - \Delta$, where Δ denotes the Laplacian operator on \mathbb{R}^d , then we impose initial conditions of the form

$$u(0,x) = u_0(x) \qquad x \in \mathbb{R}^d, \tag{4.2}$$

for some Borel function $u_0: \mathbb{R}^d \to \mathbb{R}$. If *L* is second order in time, such as the wave operator $L = \partial^{-2}/\partial t^2 - \Delta$, then we have to impose two initial conditions:

$$u(0,x) = u_0(x), \quad \frac{\partial u}{\partial t}(0,x) = v_0(x), \qquad x \in \mathbb{R}^d,$$
(4.3)

for some Borel functions $u_0, v_0 : \mathbb{R}^d \to \mathbb{R}$.

4.1. The random field approach

We now describe the notion of *mild random field* solution to Eq. (4.1). Recall that we are given a filtered probability space $(\Omega, \mathcal{F}, (\mathcal{F}_t), P)$, where $(\mathcal{F}_t)_t$ is the filtration generated by the standard cylindrical Wiener process W of Proposition 2.5, and we fix a time horizon T > 0. A real-valued adapted stochastic process $\{u(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ is a *mild random field* solution of (4.1) if the following stochastic integral equation is satisfied:

$$u(t, x) = I_0(t, x) + \int_0^t \int_{\mathbb{R}^d} \Gamma(t - s, x - y) \sigma(u(s, y)) W(ds, dy) + \int_0^t ds \int_{\mathbb{R}^d} \Gamma(s, dy) b(u(t - s, x - y)), \quad a.s.,$$
(4.4)

for all $(t, x) \in [0, T] \times \mathbb{R}^d$. In (4.4), Γ denotes the fundamental solution associated to *L* and $I_0(t, x)$ is the contribution of the initial conditions, which we define below. The stochastic integral on the

right hand-side of (4.4) is as defined in Section 2.3. In particular, we need to assume that for any (t, x), the fundamental solution $\Gamma(t - \cdot, x - \star)$ satisfies Hypothesis 2.8, and to require that $s \mapsto \Gamma(t - s, x - \star)$ $(x - \star)\sigma(u(s, \star)), s \in [0, t]$, defines a predictable process taking values in the space U of Section 2.2 such that

$$E\left(\int_0^t \|\Gamma(t-s,x-\star)\sigma(u(s,\star))\|_U^2\,ds\right) < +\infty$$

(see Sections 2.2 and 2.4). As we will make explicit in Section 4.3, these assumptions will be satisfied under certain regularity assumptions on the coefficients b and σ (see Theorem 4.3).

The last integral on the right-hand side of (4.4) is considered in the pathwise sense, and we use the notation " $\Gamma(s, dy)$ " because we will assume that $\Gamma(s)$ is a measure on \mathbb{R}^d . Concerning the term $I_0(t, t)$ x), if L is a parabolic-type operator and we consider the initial condition (4.2), then

$$I_0(t,x) = (\Gamma(t) * u_0)(x) = \int_{\mathbb{R}^d} u_0(x-y) \,\Gamma(t,dy).$$
(4.5)

On the other hand, in the case where *L* is second order in time with initial values (4.3),

$$I_{0}(t,x) = (\Gamma(t) * v_{0})(x) + \frac{\partial}{\partial t} (\Gamma(t) * u_{0})(x)$$

=
$$\int_{\mathbb{R}^{d}} v_{0}(x-y) \Gamma(t,dy) + \frac{\partial}{\partial t} \left(\int_{\mathbb{R}^{d}} u_{0}(x-y) \Gamma(t,dy) \right).$$
(4.6)

4.2. Examples: stochastic heat and wave equations

In the case of the stochastic heat equation in any space dimension d > 1 and the stochastic wave equation in dimensions d = 1, 2, 3, following [6, Section 3] (see also [29, Examples 4.2 and 4.3]), the fundamental solutions are well-known and the conditions in Hypothesis 2.8 can be made explicit. Indeed, let Γ be the fundamental solution of the heat equation in \mathbb{R}^d , d > 1, so that

$$\Gamma(t, x) = (4\pi t)^{-d/2} \exp\left(-\frac{|x|^2}{4t}\right).$$

In particular, we have $\mathcal{F}\Gamma(t)(\xi) = \exp(-4\pi^2 t |\xi|^2), \xi \in \mathbb{R}^d$, and, because

$$\int_0^1 \exp(-4\pi^2 t |\xi|^2) \, dt = \frac{1}{4\pi^2 |\xi|^2} (1 - \exp(-4\pi^2 T |\xi|^2)),$$

we conclude that condition (2.11) in Hypothesis 2.8 holds if and only if

$$\int_{\mathbb{R}^d} \frac{\mu(d\xi)}{1+|\xi|^2} < +\infty.$$
(4.7)

Now let Γ_d be the fundamental solution of the wave equation in \mathbb{R}^d , with d = 1, 2, 3. This restriction on the space dimension is due to the fact that the fundamental solution in \mathbb{R}^d with d > 3 is no longer a non-negative distribution (for results on the stochastic wave equation in spatial dimension d > 3, we refer the reader to [5]: see Remark 2.10). It is well known (see [17, Chapter 5]) that

$$\Gamma_1(t,x) = \frac{1}{2} \mathbb{1}_{\{|x| < t\}}, \quad \Gamma_2(t,x) = \frac{1}{2\pi} (t^2 - |x|^2)_+^{-1/2}, \quad \Gamma_3(t)(dx) = \frac{1}{4\pi t} \sigma_t(dx),$$

where σ_t denotes the uniform surface measure on the three-dimensional sphere of radius t, with total mass $4\pi t^2$. This implies that, for each t, $\Gamma_d(t)$ has compact support. Furthermore, for all dimensions $d \ge 1$, the Fourier transform of $\Gamma_d(t)$ is

$$\mathcal{F}\Gamma_d(t)(\xi) = \frac{\sin(2\pi t|\xi|)}{2\pi |\xi|}.$$

Elementary estimates show that there are positive constants c_1 and c_2 depending on T > 0 such that

$$\frac{c_1}{1+|\xi|^2} \le \int_0^T \frac{\sin^2(2\pi t |\xi|)}{4\pi^2 |\xi|^2} dt \le \frac{c_2}{1+|\xi|^2}$$

Therefore, Γ_d satisfies condition (2.11) if and only if (4.7) holds.

For d = 1, $I_0(t, x)$ is given by the so-called d'Alembert's formula (see, for instance, [16, p. 68]):

$$I_0^1(t,x) = \frac{1}{2} \left[u_0(x+t) + u_0(x-t) \right] + \frac{1}{2} \int_{x-t}^{x+t} v_0(y) \, dy, \qquad x \in \mathbb{R}.$$
(4.8)

For d = 2 (see [16, p. 74]),

$$I_0^2(t,x) = \frac{1}{2\pi t} \int_{|x-y| < t} \frac{u_0(y+tv_0) + \nabla u_0(y) \cdot (x-y)}{(t^2 - |x-y|^2)^{1/2}} \, dy, \qquad x \in \mathbb{R}^2.$$

Finally, for d = 3 (see [16, p. 77]), for $x \in \mathbb{R}^3$,

$$I_0^3(t,x) = \frac{1}{4\pi t^2} \int_{\mathbb{R}^3} \left(t v_0(x-y) + u_0(x-y) + \nabla u_0(x-y) \cdot y \right) \, \sigma_t(dy). \tag{4.9}$$

It is important to remark that in the above formulas, we have implicitly assumed that all integrals that appear are well defined. Indeed, in Lemma 4.2 below, we will exhibit sufficient conditions on u_0 and v_0 under which such integrals exist and are uniformly bounded with respect to t and x.

4.3. Random field solutions with arbitrary initial conditions

The aim of this section is to prove the existence and uniqueness of a mild random field solution to the stochastic integral equation (4.4).

We are interested in solutions that are L^p -bounded, as in (4.10) below, and L^2 -continuous. This is only possible under certain assumptions on the initial conditions. In particular, the initial conditions will have to be such that the following hypothesis is satisfied.

Hypothesis 4.1. $(t, x) \mapsto I_0(t, x)$ is continuous and $\sup_{(t, x) \in [0, T] \times \mathbb{R}^d} |I_0(t, x)| < +\infty$.

For the particular case of the heat equation in any spatial dimension and the wave equation with $d \in \{1, 2, 3\}$, sufficient conditions for Hypothesis 4.1 to hold are given in the next lemma.

Lemma 4.2. Consider the following two sets of hypotheses:

- (i) Heat equation. $u_0 : \mathbb{R}^d \to \mathbb{R}$ is measurable and bounded.
- (ii) Wave equation. When $d = 1, u_0$ is bounded and continuous, and v_0 is bounded and measurable. When $d = 2, u_0 \in C^1(\mathbb{R}^2)$ and there is $q_0 \in [2, \infty]$ such that $u_0, \nabla u_0, v_0$ all belong to $L^{q_0}(\mathbb{R}^2)$. When d = 3, $u_0 \in C^1(\mathbb{R}^3)$, u_0 and ∇u_0 are bounded, and v_0 is bounded and continuous.

Then under condition (i) or (ii), Hypothesis 4.1 is satisfied.

Proof. Assume first that *L* is the heat operator on \mathbb{R}^d , $d \ge 1$, with initial condition u_0 satisfying (i). Then, by (4.5),

$$\sup_{(t,x)\in[0,T]\times\mathbb{R}^d}|I_0(t,x)|\leq \|u_0\|_{\infty}\sup_{t\in [0,T[}\int_{\mathbb{R}^d}(2\pi t)^{-d/2}\exp\left(-\frac{|y|^2}{2t}\right)dy=\|u_0\|_{\infty}<+\infty.$$

Secondly, assume that *L* is the wave operator on \mathbb{R}^d , d = 1, 2, 3, and that condition (ii) is satisfied. We make explicit the dependence on the space dimension by denoting $I_0^d(t, x)$, d = 1, 2, 3, the term $I_0(t, x)$.

By (4.8), if d = 1 it is clear that

$$\sup_{(t,x)\in[0,T]\times\mathbb{R}}|I_0^1(t,x)|\leq C(\|u_0\|_{\infty}+\|v_0\|_{\infty}).$$

94

To deal with the case d = 2, we refer to [26, pp. 808–809]. In this reference, the explicit formula $\Gamma_2(t, x) = \frac{1}{2\pi} (t^2 - |x|^2)_+^{-1/2}$ was used to show that

$$\sup_{(t,x)\in[0,T]\times\mathbb{R}}|I_0^2(t,x)|\leq C(\|u_0\|_{\infty}+\|\nabla u_0\|_{\infty}+\|v_0\|_{\infty}).$$

Finally, for the case d = 3 we have, by (4.9):

$$|l_0^3(t,x)| \le C(||v_0||_{\infty} + ||u_0||_{\infty} + ||\nabla u_0||_{\infty}) \sup_{s \in]0, T[} \frac{\sigma_s(\mathbb{R}^3)}{s^2},$$

where σ_s denotes the uniform surface measure on the three-dimensional sphere of radius *s*. In particular, the total mass of σ_s is proportional to s^2 and, therefore, $I_0^3(t, x)$ is uniformly bounded with respect to $t \in [0, T]$ and $x \in \mathbb{R}^3$.

(3)

Finally, the continuity property of $(t, x) \mapsto I_0(t, x)$ follows from the hypotheses and the explicit formulas for $I_0(t, x)$ given in Section 4.1. This concludes the proof. \Box

The next theorem discusses existence and uniqueness of mild random field solutions to equation (4.4). Since this theorem covers rather general initial conditions, it is an extension of Theorem 13 in [6]. Indeed, in this reference, only vanishing initial data could be considered, because of the spatially homogeneous covariance required for the process *Z* in the construction of the stochastic integral used there for the wave equation when d = 3 (see [6, p. 10 and Theorem 2]). Of course, in the case of the stochastic wave equation in spatial dimensions d = 1, 2, there are many results on existence and uniqueness of mild random field solutions with non-vanishing initial conditions: see for instance [3,7,26,28].

Theorem 4.3. Assume that Hypotheses 2.8 and 4.1 are satisfied and that σ and b are Lipschitz functions. Then there exists a unique mild random field solution $\{u(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ of Eq. (4.4). Moreover, the process u is L^2 -continuous and for all $p \ge 1$,

$$\sup_{\substack{(t,x)\in[0,T]\times\mathbb{R}^d}} E(|u(t,x)|^p) < +\infty.$$
(4.10)

Proof. The proof is similar to those of [26, Theorem 1.2] and [6, Theorem 13]. We define the Picard iteration scheme

$$u^{0}(t,x) = I_{0}(t,x),$$

$$u^{n+1}(t,x) = u^{0}(t,x) + \int_{0}^{t} \int_{\mathbb{R}^{d}} \Gamma(t-s,x-y)\sigma(u^{n}(s,y)) W(ds,dy) + \int_{0}^{t} \int_{\mathbb{R}^{d}} b(u^{n}(t-s,x-y)) \Gamma(s,dy) ds,$$
(4.11)

for $n \ge 0$. We prove by induction on n that the process $\{u^n(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ is well defined and, for $p \ge 1$,

$$\sup_{(t,x)\in[0,T]\times\mathbb{R}^d} E(|u^n(t,x)|^p) < +\infty,$$
(4.12)

for every $n \ge 0$.

(t,

Notice that by Hypothesis 4.1, the process u^0 is locally bounded, and the Lipschitz property on σ yields

$$\sup_{x)\in[0,T]\times\mathbb{R}^d}|\sigma(u^0(t,x))|^p<+\infty.$$

By Proposition 2.9, this implies that the stochastic integral

$$\mathcal{I}^{0}(t,x) = \int_{0}^{t} \int_{\mathbb{R}^{d}} \Gamma(t-s,x-y)\sigma(u^{0}(s,y)) W(ds,dy)$$

is well-defined and

$$E(|\mathcal{I}^{0}(t,x)|^{p}) \leq C \int_{0}^{t} ds \sup_{z \in \mathbb{R}^{d}} (1 + |u^{0}(s,z)|^{p}) \int_{\mathbb{R}^{d}} \mu(d\xi) |\mathcal{F}\Gamma(t-s)(\xi)|^{2}$$

$$\leq C \sup_{(s,z)\in[0,T]\times\mathbb{R}^{d}} (1 + |u^{0}(s,z)|^{p}) \int_{0}^{T} ds J(s), \qquad (4.13)$$

where

$$J(s) = \int_{\mathbb{R}^d} \mu(d\xi) |\mathcal{F}\Gamma(s)(\xi)|^2.$$

In order to deal with the pathwise integral

$$\mathcal{J}^0(t,x) = \int_0^t ds \int_{\mathbb{R}^d} \Gamma(s,dy) b(u^0(t-s,x-y)),$$

we apply Hölder's inequality with respect to the finite measure $\Gamma(s, dy) ds$ on $[0, T] \times \mathbb{R}^d$ and use the Lipschitz property of *b*:

$$\left|\mathcal{J}^{0}(t,x)\right|^{p} \leq C \int_{0}^{t} ds \int_{\mathbb{R}^{d}} \Gamma(s,dy) \left(1 + \left|u^{0}(t-s,x-y)\right|^{p}\right).$$
(4.14)

The latter term is uniformly bounded with respect to *t* and *x*. Together with (4.13), this implies that $\{u^1(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ is a well-defined measurable process. Further, by (4.13), (4.14) and Hypothesis 2.8,

$$\sup_{(t,x)\in[0,T]\times\mathbb{R}^d} E(|u^1(t,x)|^p) < +\infty.$$

Consider now n > 1 and assume that $\{u^n(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ is a well-defined measurable process satisfying (4.12). Using the same arguments as above, one proves that the integrals $\mathcal{I}^{n+1}(t, x)$ and $\mathcal{J}^{n+1}(t, x)$ exist, so that the process u^{n+1} is well-defined and is uniformly bounded in $L^p(\Omega)$. This proves (4.12).

The next step consists in showing that the bound (4.12) is uniform with respect to *n*, that is

$$\sup_{n \ge 0} \sup_{(t,x) \in [0,T] \times \mathbb{R}^d} E(|u^n(t,x)|^p) < +\infty.$$
(4.15)

Indeed, the same kind of estimates as in the first part of the proof show that for $n \ge 1$,

$$E(|u^{n+1}(t,x)|^p) \le C\left(1 + \int_0^t ds\left(1 + \sup_{z \in \mathbb{R}^d} E(|u^n(s,z)|^p)\right) (J(t-s)+1)\right).$$

We conclude that (4.15) holds by the version of Gronwall's Lemma presented in [6, Lemma 15].

Now we show that the sequence $(u^n(t, x))_{n \ge 1}$ converges in $L^p(\Omega)$. Following the same lines as in the proof of [6, Theorem 13], let

$$M_n(t) := \sup_{(s,x)\in[0,t]\times\mathbb{R}^d} E(|u^{n+1}(s,x) - u^n(s,x)|^p)$$

Using the Lipschitz property of b and σ , and applying the same arguments as above, we obtain the estimate

$$M_n(t) \le C \int_0^t ds M_{n-1}(s) (J(t-s)+1).$$

Hence, we apply again [6, Lemma 15] to conclude that $(u^n(t, x))_{n\geq 1}$ converges uniformly in $L^p(\Omega)$ to a limit u(t, x). The process $\{u(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ has a measurable version that satisfies Eq. (4.4).

96

Indeed, let us sketch the calculations concerning the stochastic integral term $\mathcal{I}^n(t, x)$ of (4.11): we will prove that

$$\lim_{n\to\infty} \sup_{(t,x)\in[0,T]\times\mathbb{R}^d} E(|\mathcal{I}^n(t,x)-\mathcal{I}(t,x)|^p) = 0.$$

By the Lipschitz property of σ , Proposition 2.9 and Hypothesis 2.8,

$$E(|\mathcal{I}^{n}(t,x) - \mathcal{I}(t,x)|^{p}) \leq E\left(\left|\int_{0}^{t} \int_{\mathbb{R}^{d}} \Gamma(t-s,x-y)[\sigma(u^{n-1}(s,y)) - \sigma(u(s,y))]W(ds,dy)\right|^{p}\right)$$

$$\leq C \int_{0}^{t} ds \sup_{z \in \mathbb{R}^{d}} E(|u^{n-1}(s,z) - u(s,z)|^{p}) \int_{\mathbb{R}^{d}} \mu(d\xi)|\mathcal{F}\Gamma(t-s)(\xi)|^{2}$$

$$\leq C \sup_{(s,z)\in[0,T]\times\mathbb{R}^{d}} E(|u^{n-1}(s,z) - u(s,z)|^{p})$$

and this last term converges to zero, as n tends to infinity. The pathwise integral term can be studied in a similar manner. Therefore, the process u solves (4.4). Finally, uniqueness of the solution can be checked by standard arguments. \Box

4.4. Spatially homogeneous spde's in the infinite-dimensional setting

Stochastic partial differential equations of the form (4.1) on \mathbb{R}^d and driven by a spatially homogeneous Wiener process have been studied, in the context of Da Prato and Zabczyk [14], in a series of works: [15, Section 11.4], and [20,21,30–32]. The aim of this section is to sketch the formulation used in those papers, focusing mostly on the one used by Peszat and Zabczyk in [32]. Then, in Section 4.5, we will compare their solution with the mild random field solution of Section 4.1.

In [32], the stochastic wave equation with d = 1, 2, 3 and the stochastic heat equation in any space dimension are considered. This meshes well with the case considered in Section 4.3. However, we note that the stochastic wave equation in higher dimensions (d > 3) can also be formulated and solved in the infinite-dimensional setting, but using a slightly different formulation (see [30]).

4.4.1. General framework

We first recall the generic setup for evolution equations in infinite-dimensional spaces. These are usually written

$$\begin{cases} du(t) = (Au(t) + F(u(t))) dt + B(u(t)) d\mathcal{W}_t, & t \in]0, T[, \\ u(0) = h. \end{cases}$$
(4.16)

In this equation, (W_t) is a cylindrical *Q*-Wiener process on a Hilbert space *V*, *h* is an element of a Hilbert space *H*, *F* is a mapping from *H* into *H*, and *B* is a mapping from *H* into L(U, H). The operator $A: D(A) \subset H \to H$ is the infinitesimal generator of a (strongly) continuous semigroup $(S(t))_{t \in \mathbb{R}_+}$ (meaning, generically, that $S(0) = I_V$, S(t + s) = S(t)S(s), and for $h \in D(A)$, $\frac{d}{dt}S(t)h = AS(t)h = S(t)Ah$, so that one sometimes writes $S(t) = e^{At}$).

The process (W_t) is assumed to be adapted to a filtration $\{\mathcal{F}_t, t \in \mathbb{R}_+\}$, such that for all $s, t \in \mathbb{R}_+$, $W_{t+s} - W_t$ is independent of \mathcal{F}_t .

An adapted *H*-valued process $\{u(t), t \in [0, T]\}$ is a *mild* solution of (4.16) provided for all $t \in [0, T]$, a.s.,

$$u(t) = S(t)h + \int_0^t S(t-s)F(u(s)) \, ds + \int_0^t S(t-s)B(u(s)) \, d\mathcal{W}_s.$$
(4.17)

Of course, (*S*(*t*)), *F*, *B* and (W_t) must satisfy certain conditions in order that this equation make sense, and further conditions in order to guarantee existence and uniqueness of the solution. It is also necessary to specify the meaning of the two integrals in (4.17): the first is essentially a Bochner

integral, and for the second, one assumes that for all $h \in H$ and $t \in \mathbb{R}_+$, S(t)B(h) belongs to $L_2(V_0, H)$, where $V_0 = Q^{1/2}(V)$. For existence and uniqueness, F and B must typically satisfy at least Lipschitz and linear growth conditions (see, for instance, [15, Theorem 5.3.1]).

In order to place the spde (4.1) in this framework, it is necessary to specify all the ingredients above (V, W, H, A, S, F, and B). We begin with V and W.

4.4.2. Spatially homogeneous noise

To begin with, we note that in [32]—and, indeed, in the above mentioned companion papers—a slightly more general spatially correlated noise than the one described in Section 2.2 has been used. More precisely, one considers a spatially homogeneous Wiener process $\{\mathcal{W}_t^*, t \ge 0\}$ with values in the space $\mathcal{S}'(\mathbb{R}^d)$ of tempered distributions. Denoting by $\langle \cdot, \cdot \rangle$ the usual duality action of $\mathcal{S}'(\mathbb{R}^d)$ on $\mathcal{S}(\mathbb{R}^d)$, this means that for all $\varphi \in \mathcal{S}(\mathbb{R}^d)$, $\{\langle \mathcal{W}_t^*, \varphi \rangle, t \in \mathbb{R}_+\}$ is a centered Gaussian process and there exists $\Lambda \in \mathcal{S}'(\mathbb{R}^d)$ such that for all $\varphi, \psi \in \mathcal{S}(\mathbb{R}^d)$ and $s, t \in \mathbb{R}_+$,

$$E\left(\left\langle \mathcal{W}_{s}^{*},\varphi\right\rangle\left\langle \mathcal{W}_{t}^{*},\psi\right\rangle\right)=(s\wedge t)\langle\Lambda,\varphi*\tilde{\psi}\rangle,$$

where $\tilde{\psi}(x) = \psi(-x)$. The Schwartz distribution Λ must be the Fourier transform of a symmetric and non-negative tempered measure μ on \mathbb{R}^d .

Remark 4.4. In the particular case where Λ is a non-negative measure satisfying the conditions of Section 2.2, we recover the covariance operator of the cylindrical Wiener process W on the Hilbert space U defined in Proposition 2.5 (see (2.2)):

$$E\left(\left\langle \mathcal{W}_{s}^{*},\varphi\right\rangle\left\langle \mathcal{W}_{t}^{*},\psi\right\rangle\right)=(s\wedge t)\int_{\mathbb{R}^{d}}\Lambda(dx)\left(\varphi*\tilde{\psi}\right)(x)=E\left(W_{s}(\varphi)W_{t}(\psi)\right).$$
(4.18)

In order to relate this general noise to the one defined in Section 2.2, let U be the Hilbert space defined in Section 2.2, and let U^* be the dual of U. The following characterization of U^* is given in [31, Proposition 1.2]. Recall that $\tilde{L}^2(\mathbb{R}^d, \mu)$ stands for the subspace of $L^2(\mathbb{R}^d, \mu)$ consisting of all functions ϕ such that $\tilde{\phi} = \phi$.

Lemma 4.5. A distribution $g \in S'(\mathbb{R}^d)$ belongs to U^* if and only if there is $\phi \in \tilde{L}^2(\mathbb{R}^d, \mu)$ such that $g = \mathcal{F}(\phi_1\mu)$. Moreover, if $g_1 = \mathcal{F}(\phi_1\mu)$ and $g_2 = \mathcal{F}(\phi_2\mu)$, with $\phi_1, \phi_2 \in \tilde{L}^2(\mathbb{R}^d, \mu)$, then

$$\langle g_1, g_2 \rangle_{U^*} = \langle \phi_1, \phi_2 \rangle_{\tilde{L}^2(\mathbb{R}^d, \mu)}.$$

Remark 4.6. The previous lemma allows us to determine the explicit form of the isometry $I : U \to U^*$. More precisely, as stated in Remark 2.3, any element $g \in U$ can be written in the form $g = \mathcal{F}^{-1}\phi$, with $\phi \in \tilde{L}^2(\mathbb{R}^d; d\mu)$. Then, for such $g, I(g) \in U^*$ is defined by

$$I(\mathbf{g}) = \mathcal{F}(\boldsymbol{\phi}\boldsymbol{\mu}).$$

Moreover, we have the following lemma whose proof is straightforward. In this lemma, $\tilde{S}(\mathbb{R}^d)$ denotes the family of functions $\varphi \in S(\mathbb{R}^d)$ such that $\tilde{\varphi} = \varphi$.

Lemma 4.7. Let $\varphi \in U$ be such that $\varphi \in \tilde{S}(\mathbb{R}^d)$. Then $I(\varphi) = \varphi * \Lambda$.

As it has been explained in [31, p. 191] (see, in particular, Proposition 1.1 therein), W^* may be regarded as a U^* -valued cylindrical Q-Wiener process with $Q = Id_{U^*}$ (so that the generic Hilbert space V used in (4.16) is $V = U^*$). More precisely, let U_1^* be a Hilbert space such that there exists a dense Hilbert–Schmidt embedding $J^* : U^* \to U_1^*$ (see Proposition 3.6). Then

$$\mathcal{W}_{t}^{*} = \sum_{j=1}^{\infty} \beta_{j}(t) J^{*}(e_{j}^{*}), \tag{4.19}$$

where $(e_j^*)_j$ is a complete orthonormal basis in U^* , and the $\beta_j(t)$ are independent standard Brownian motions (note that $Q^{1/2} = Q^{-1/2} = \text{Id}_{U^*}$). Therefore, we will be able to define Hilbert-space-valued

stochastic integrals with respect to W^* , as has been described in Section 3.5. Note that U^* is sometimes called the reproducing kernel Hilbert space associated to W^* (see, for instance, [14, Section 2.2.2] or [31, p. 191]).

4.4.3. The space H and the operators A, S, F and B

In [32], mild solutions to the formal Eq. (4.1) are considered in a Hilbert space H of the form $L^2_{\vartheta} = L^2(\mathbb{R}^d, \vartheta(x)dx)$, where $\vartheta \in C^{\infty}(\mathbb{R}^d)$ is a strictly positive even function such that $\vartheta(x) = e^{-|x|}$, for $|x| \ge 1$. Let us also denote by H^1_{ϑ} the weighted Sobolev space which is the completion of $\mathcal{S}(\mathbb{R}^d)$ with respect to the norm

$$\|\psi\|_{H^1_{\vartheta}} = \left(\int_{\mathbb{R}^d} \left[|\psi(x)|^2 + |\nabla\psi(x)|^2\right] \vartheta(x) dx\right)^{1/2}.$$

The operator *A* is the Laplacian Δ , with domain the classical Sobolev space $H^{2,2}$. In the case of the heat equation, where $L = d/dt - \Delta$, the associated semigroup S(t) is such that

$$S(t)\varphi(x) = \int_{\mathbb{R}^d} \Gamma(t, x - y)\varphi(y) \, dy, \quad \varphi \in \mathcal{S}(\mathbb{R}^d), \quad t \ge 0, \quad x \in \mathbb{R}^d,$$
(4.20)

where Γ is the fundamental solution of the heat equation in \mathbb{R}^d (see Section 4.2), and it is shown in [31, Lemma 3.1] that *S* has a unique extension to a (holomorphic) semigroup on L^2_{ϑ} , which we still denote by *S*.

The operators *F* and *B* are the so-called *Nemitskii operators* associated respectively to the functions *b* and σ appearing in (4.1), and are defined by

$$F(v)(x) := b(v(x)), \qquad B(v)(x) := \sigma(v(x)), \qquad v \in L^2_{\vartheta}, \quad x \in \mathbb{R}^d$$

With the choices just made, the mild formulation (4.17), with W replaced there by W^* , is a formalization of (4.1) in the case where $L = \frac{d}{dt} - \Delta$ (the stochastic heat equation).

4.4.4. The case of the stochastic wave equation

In the case of the wave equation, $L = \partial^2/\partial t^2 - \Delta$, the formulations (4.16) and (4.17) are not immediately applicable, because *L* is not first-order in time. At least two approaches are possible: the second-order equation (in time) can be written as a system of first-order equations, or one can focus on the mild formulation (4.17), with *S*(*t*) defined by

$$S(t)\varphi = \Gamma(t) * \varphi,$$

where $\Gamma(t)$ is now the fundamental solution of the wave equation, as in Section 4.2. In this case, S(t) no longer defines a semigroup. Nevertheless, this approach was used in [32] in order to treat the heat and wave equations in a unified manner.

When $L = \frac{\partial^2}{\partial t^2} - \Delta$, we will restrict ourselves to spatial dimensions $d \in \{1, 2, 3\}$. Let Γ be the fundamental solution associated to L, let $u_0 \in H^1_{\vartheta}$, $v_0 \in L^2_{\vartheta}$, and fix a time horizon T > 0. By definition, a *mild* L^2_{ϑ} -valued solution of (4.1) with $L = \frac{\partial^2}{\partial t^2} - \Delta$, is an \mathcal{F}_t -adapted process $\{u(t), t \in [0, T]\}$ with values in L^2_{ϑ} satisfying

$$u(t) = \frac{\partial}{\partial t} \left(\Gamma(t) * u_0 \right) + \Gamma(t) * v_0 + \int_0^t \Gamma(t-s) * b(u(s)) \, ds + \int_0^t \Gamma(t-s) * \sigma(u(s)) \, d\mathcal{W}_s^*.$$
(4.21)

The stochastic integral on the right-hand side of (4.21) has to be defined. This requires interpreting the integrand $\Gamma(t - s) * \sigma(u(s))$ in the framework of Section 3.5.

Recall that, as in Section 3.5 and since $Q = Id_{U^*}$ and so $U^* = (U^*)_0$, we will be able to define the stochastic integral with respect to W^* of any predictable process Φ taking values in the space $L_2(U^*, H)$, where $H = L_{\vartheta}^2$. Therefore, it is necessary to interpret $\Gamma(t - s) * \sigma(u(s))$ as an element of $L_2(U^*, H)$.

Let $U^{*,0}$ be the dense subspace of U^* consisting of all $g = \mathcal{F}(\phi\mu)$ with $\phi \in \tilde{\mathcal{S}}(\mathbb{R}^d)$. According to [32, p. 427], it holds that $U^{*,0} \subset C_b(\mathbb{R}^d)$, the space of bounded and continuous functions on \mathbb{R}^d . For $u \in L^2_{\vartheta}$ and t > 0, define the following operator:

$$\mathcal{K}(t,u)(\eta) = \Gamma(t) * (u\eta), \qquad \eta \in U^{*,0}.$$
(4.22)

Then it is shown in Lemma 3.3 of [32] that, for all t > 0 and $u \in L^2_{\vartheta}$, $\mathcal{K}(t, u)$ has a unique extension to a Hilbert–Schmidt operator from U^* into L^2_{ϑ} . Thus extended, $\mathcal{K}(t, \cdot)$ becomes a bounded linear operator from L^2_{ϑ} into $L_2(U^*, L^2_{\vartheta})$. Therefore, if u is an L^2_{ϑ} -valued adapted process, we can define the stochastic integral as follows:

$$\int_{0}^{t} \left(\Gamma(t-s) * \sigma(u(s)) \right) \, d\mathcal{W}_{s}^{*} := \int_{0}^{t} \mathcal{K}(t-s, \sigma(u(s))) \, d\mathcal{W}_{s}^{*}.$$
(4.23)

In the formulation above, $\sigma(u(s))$ denotes the function $\sigma(u(s))(x) := \sigma(u(s, x)), x \in \mathbb{R}^d$, which belongs to L^2_{ϑ} .

This definition of the stochastic integral (4.23) is the one that is used in the mild formulation (4.21). The main result in [32] on existence and uniqueness of a solution to Eq. (4.21) is the following (see [32, Theorem 0.1]).

Theorem 4.8. Assume that $d \in \{1, 2, 3\}$ and that the coefficients *b* and σ are Lipschitz functions. Suppose that there is $\kappa > 0$ such that $\Lambda + \kappa$ dx is a nonnegative measure (where dx denotes Lebesgue measure), and the spectral measure μ satisfies

$$\int_{\mathbb{R}^d} \frac{\mu(d\xi)}{1+|\xi|^2} < +\infty.$$

$$(4.24)$$

Then, for arbitrary $u_0 \in H^1_{\vartheta}$ and $v_0 \in L^2_{\vartheta}$, there exists a unique L^2_{ϑ} -valued solution to Eq. (4.21).

4.4.5. The stochastic heat equation

We now return to the case of the stochastic heat equation, namely we consider Eq. (4.1) when $L = \partial/\partial t - \Delta$, with any spatial dimension $d \ge 1$.

Letting Γ be the fundamental solution of the heat equation in \mathbb{R}^d , and defining *S* as in (4.20), the formulation in (4.17) is equivalent (see [32, Section 5]) to the equation

$$u(t) = \Gamma(t) * u_0 + \int_0^t \Gamma(t-s) * b(u(s)) \, ds + \int_0^t \Gamma(t-s) * \sigma(u(s)) \, d\mathcal{W}_s^*.$$
(4.25)

Similar to (4.23), the stochastic integral here is defined by

$$\int_0^t \Gamma(t-s) * \sigma(u(s)) \, d\mathcal{W}_s^* := \int_0^t \mathcal{P}(t-s, \sigma(u(s))) \, d\mathcal{W}_s^*,$$

where the operator $\mathcal{P}(t, u)$, for $u \in L^2_{\vartheta}$ and t > 0, is defined as in (4.22), but Γ is now the fundamental solution of the heat equation in \mathbb{R}^d . As explained in [32, Section 5], the equivalence between (4.25) and (4.17) can be understood through the equality

$$\mathcal{P}(t, u)(\eta) = S(t)(u\eta), \qquad u \in L^2_{\vartheta}, \quad \eta \in U^{*, 0}.$$

Moreover, [32, Lemma 5.3] states that $\mathcal{P}(\cdot, u)$ defines a square-integrable process with values in $L_2(U^*, L_{\vartheta}^2)$, which implies that the above stochastic integral is well-defined.

The main result in [32] on existence and uniqueness of a solution to Eq. (4.25) is the following (see [32, Theorem 0.2]).

Theorem 4.9. Assume that $d \ge 1$ and that the coefficients b and σ are Lipschitz functions. Suppose that there is $\kappa > 0$ such that $\Lambda + \kappa dx$ is a nonnegative measure, and the spectral measure μ satisfies (4.24). Then, for arbitrary $u_0 \in L^2_{\vartheta}$, there exists a unique L^2_{ϑ} -valued solution to Eq. (4.25).

100

4.5. Relation with the random field approach

We now examine the relationship between the random field solution to equation (4.4) in the case of the stochastic wave and heat equations and the L^2_{ϑ} -valued solution to equations (4.21) and (4.25), respectively. For this, we assume that the cylindrical Wiener process W considered in the beginning of Section 4.1 and the cylindrical Q-Wiener process W^* (with $Q = Id_{U^*}$) that appears in (4.21) are related as follows.

Let $(e_j)_j$ be a complete orthonormal basis of the Hilbert space U such that $e_j \in \tilde{S}(\mathbb{R}^d)$, for all $j \ge 1$. Assume that the e_i^* and the $\beta_j(t)$ that appear in (4.19) are given by

$$e_j^* = I(e_j)$$
 and $\beta_j(t) = W_t(e_j)$, (4.26)

where *I* is the isometry described in Remark 4.6 (in particular, (e_j^*) is the dual basis of (e_j) , that is, $\langle e_j^*, e_k \rangle_{U^*, U} = \delta_{j, k} \rangle$. Recall that $J^* : U^* \to U_1^*$ denotes a Hilbert–Schmidt embedding between U^* and a possibly larger Hilbert space U_1^* ; moreover, by Proposition 3.6, $(J^*(e_i^*))_j$ defines a basis in U_1^* .

Let us consider $\{u(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$, the mild random field solution of (4.4) as given in Theorem 4.3, in the case where *L* is either the wave operator in spatial dimension $d \in \{1, 2, 3\}$ or the heat operator with $d \ge 1$ (so as to have a specific form for $I_0(t, x)$). Then for all $(t, x) \in [0, T] \times \mathbb{R}^d$,

$$u(t,x) = I_0(t,x) + \int_0^t \int_{\mathbb{R}^d} \Gamma(t-s,x-y)\sigma(u(s,y)) W(ds,dy) + \int_0^t \int_{\mathbb{R}^d} b(u(t-s,x-y)) \Gamma(s,dy) \, ds, \quad a.s.,$$
(4.27)

Here, the expression for $I_0(t, x)$ is given either by (4.5) or (4.6), and we assume that the initial conditions satisfy the hypotheses specified in Lemma 4.2. The coefficients σ and b are Lipschitz functions. Recall that $(t, x) \mapsto u(t, x)$ is mean-square continuous and satisfies

$$\sup_{(t,x)\in[0,T]\times\mathbb{R}^d} E(|u(t,x)|^2) < +\infty.$$
(4.28)

This section is devoted to proving the following result.

Proposition 4.10. Let $\{u(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ be the mild random field solution of (4.27), where *L* is either the wave operator in spatial dimension $d \in \{1, 2, 3\}$ or the heat operator with $d \ge 1$. Let $u(t) = u(t, \star)$. Then $\{u(t), t \in [0, T]\}$ is the mild L^2_{θ} -valued solution of (4.21) or (4.25), respectively.

Proof. This proof is written for the case of the stochastic wave equation in spatial dimension $d \in \{1, 2, 3\}$, but also applies to the stochastic heat equation with $d \ge 1$.

In view of the integral Eqs. (4.27) and (4.21), it is clear that the most delicate part in the proof corresponds to the analysis of the stochastic integral terms. Hence, we will start by assuming that both the initial conditions and the drift term *b* vanish. In this case, $\{u(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ solves the integral equation

$$u(t,x) = \int_0^t \int_{\mathbb{R}^d} \Gamma(t-s,x-y)\sigma(u(s,y)) W(ds,dy), \quad a.s$$

for all $(t, x) \in [0, T] \times \mathbb{R}^d$. Let us use the following notation for the above stochastic integral:

$$\mathcal{I}(t,x) := \int_0^t \int_{\mathbb{R}^d} \Gamma(t-s,x-y)\sigma(u(s,y)) W(ds,dy).$$

For any $(t, x) \in [0, T] \times \mathbb{R}^d$, the above integral is a real-valued random variable and it is well-defined because the integrand satisfies the hypotheses described in Section 2.4, that is, $\Gamma(t - \cdot, x - \star)$

verifies Hypothesis 2.8 and { $\sigma(u(s, y))$, $(s, y) \in [0, t] \times \mathbb{R}^d$ } is a predictable process such that

$$\sup_{(s,y)\in[0,T]\times\mathbb{R}^d} E(|\sigma(u(s,y))|^2) \le C\left(1 + \sup_{(s,y)\in[0,T]\times\mathbb{R}^d} E(|u(s,y)|^2)\right) < +\infty.$$
(4.29)

Let $u(t) = u(t, \star)$, $t \in [0, T]$. We aim to prove that $\{u(t), t \in [0, T]\}$ defines a square-integrable stochastic process with values in the weighted space L^2_{ϑ} which satisfies

$$u(t) = \int_0^t \Gamma(t-s) * \sigma(u(s)) d\mathcal{W}_s^*, \qquad t \in [0,T].$$

Hence, our objective is to prove that $\{\mathcal{I}(t, \star), t \in [0, T]\}$ defines an element in $L^2(\Omega \times [0, T]; L^2_{\vartheta})$ and

$$\mathcal{I}(t,\star) = \int_0^t \Gamma(t-s) * \sigma(u(s)) \, d\mathcal{W}_s^*.$$

In order to simplify the notation, we will write $Z(s, y) := \sigma(u(s, y))$ and let Z(s) denote the function $Z(s)(y) = Z(s, y), y \in \mathbb{R}^d$.

We will split the proof into several steps.

Step 1. We shall check that $\{\mathcal{I}(t, \star), t \in [0, T]\}$ belongs to $L^2(\Omega \times [0, T]; L^2_{\vartheta})$ and that, for any fixed $(t, x) \in [0, T] \times \mathbb{R}^d$, the real-valued stochastic integral $\mathcal{I}(t, x)$ can be written as a stochastic integral with respect to a Hilbert-space-valued Wiener process.

Notice that the norm of $\mathcal{I}(\cdot, \star)$ in $L^2(\Omega \times [0, T]; L^2_{\vartheta})$ coincides with the norm of $u(\cdot)$ in the same space, and the latter is given by

$$E\left(\int_0^T dt \int_{\mathbb{R}^d} dx \,\vartheta(x)|u(t,x)|^2\right).$$

By (4.28) and the fact that ϑ is integrable over \mathbb{R}^d , this quantity is finite. In particular, we also deduce that *Z* belongs to $L^2(\Omega \times [0, T]; L^2_{\vartheta})$.

On the other hand, let us recall that $\mathcal{I}(t, x)$ is a stochastic integral with respect to the cylindrical Wiener process { $W_s(h)$, $s \in [0, T]$, $h \in U$ } (see Section 2.2) with covariance operator $Q = Id_U$ and $s \mapsto \Gamma(t - s, x - \star)Z(s)$ is a predictable process in $L^2(\Omega \times [0, T], U)$ by Proposition 2.9. Hence, by Proposition 3.10, the stochastic integral $\mathcal{I}(t, x)$ may be written as

$$\mathcal{I}(t,x) = \int_0^t \Phi_s^{t,x} d\mathcal{W}_s, \tag{4.30}$$

where similar to (3.19),

$$\mathcal{W}_t = \sum_{j=1}^{\infty} W_t(e_j) J(e_j),$$

 $J: U \to U_1$ is a Hilbert–Schmidt embedding from U into a possibly larger space U_1 (note that U_1 need not be the dual of U_1^* mentioned after (4.26)), and $\{\Phi_s^{t,x}, s \in [0, t]\}$ is the predictable and square integrable process with values in the space $L_2(U, \mathbb{R})$ of Hilbert–Schmidt operators from U into \mathbb{R} , given by

$$\Phi_s^{t,x}(h) = \langle \Gamma(t-s, x-\star)Z(s), h \rangle_U, \qquad h \in U.$$

Moreover,

$$E\left(\int_{0}^{t} \left\| \Phi_{s}^{t,x} \right\|_{L_{2}(U,\mathbb{R})}^{2} ds \right) = E\left(\int_{0}^{t} \left\| \Gamma(t-s,x-\star)Z(s) \right\|_{U}^{2} ds \right)$$

Step 2. Recall that we aim to prove that

$$\mathcal{I}(t,\star) = \int_0^t (\Gamma(t-s) * Z(s)) \, d\mathcal{W}_s^*, \quad t \in [0,T],$$
(4.31)

where this equality must be understood in $L^2(\Omega \times [0, T]; L^2_{\vartheta})$.

Let $t \in [0, T]$ and $(f_k)_k$ be a complete orthonormal basis in L^2_{ϑ} . We will find a suitable expansion of $\mathcal{I}(t, \star)$ in terms of $(f_k)_k$. Indeed, by (4.30) and since $\mathcal{I}(t, \star)$ defines a square integrable L^2_{ϑ} -valued random variable, we have the following representation:

$$\mathcal{I}(t,\star) = \sum_{k=1}^{\infty} \left[\int_{\mathbb{R}^d} dx \,\vartheta(x) \left(\int_0^t \Phi_s^{t,x} \, d\mathcal{W}_s \right) \cdot f_k(x) \right] f_k.$$
(4.32)

Then, by definition of the stochastic integral with respect to W and using representation (3.12) in Proposition 3.4 (for $H = \mathbb{R}$), for all $x \in \mathbb{R}^d$,

$$\int_{0}^{t} \Phi_{s}^{t,x} d\mathcal{W}_{s} = \int_{0}^{t} \Phi_{s}^{t,x} \circ J^{-1} d\mathcal{W}_{s} = \sum_{j=1}^{\infty} \int_{0}^{t} \Phi_{s}^{t,x} \circ J^{-1}(J(e_{j})) d\mathcal{W}_{s}(e_{j}),$$
$$= \sum_{j=1}^{\infty} \int_{0}^{t} \Phi_{s}^{t,x}(e_{j}) d\beta_{j}(s),$$
(4.33)

where we have made use of (4.26). Hence, plugging (4.33) into (4.32), we see that

$$\mathcal{I}(t,\star) = \sum_{k=1}^{\infty} \left[\int_{\mathbb{R}^d} dx \,\vartheta(x) \left(\sum_{j=1}^{\infty} \int_0^t \Phi_s^{t,x}(e_j) \,d\beta_j(s) \right) \cdot f_k(x) \right] f_k.$$
(4.34)

Step 3. We now give an analogous representation for the stochastic integral on the right-hand side of (4.31). For this, we will again apply Proposition 3.4 directly to the right-hand side of (4.31); notice that here, $H = L_{\vartheta}^2$, and the J^* in (4.19) cancels with the $(J^*)^{-1}$ in the definition of the stochastic integral. Therefore, taking (4.26) into account, we see that

$$\int_0^t \Gamma(t-s) * Z(s) \, d\mathcal{W}^* = \sum_{k=1}^\infty \left(\sum_{j=1}^\infty \int_0^t \left\langle \Gamma(t-s) * \left(Z(s)I(e_j) \right), f_k \right\rangle_{L^2_{\vartheta}} \, d\beta_j(s) \right) f_k, \tag{4.35}$$

where $(f_k)_k$ and β_j are as in Step 2. Recall that, on the left-hand side of (4.35), $\Gamma(t - s) * Z(s)$ is the formal notation for the Hilbert–Schmidt operator defined on U^* and taking values in L^2_{ϑ} such that, for any $\eta \in U^{0,*}$, $(\Gamma(t - s) * Z(s))(\eta) = \mathcal{K}(s, Z(s))(\eta) = \Gamma(t - s) * (Z(s)\eta)$.

By Lemma 4.7, $I(e_i) = e_i * \Lambda$ (because $e_i \in \tilde{S}(\mathbb{R}^d)$), so equality (4.35) can be written in the form

$$\int_0^t \Gamma(t-s) * Z(s) d\mathcal{W}^*$$

= $\sum_{k=1}^\infty \left(\sum_{j=1}^\infty \int_0^t \left(\int_{\mathbb{R}^d} dx \, \vartheta(x) \left[\Gamma(t-s) * \left(Z(s)(e_j * \Lambda) \right) \right](x) \cdot f_k(x) \right) d\beta_j(s) \right) f_k.$

Applying Fubini's Theorem and comparing the latter expression with (4.34), we observe that, in order to prove (4.31), it suffices to check that, for almost all $x \in \mathbb{R}^d$ and any $\varphi \in \tilde{S}(\mathbb{R}^d)$,

$$\Phi_s^{t,x}(\varphi) = \left[\Gamma(t-s) * (Z(s)(\varphi * \Lambda))\right](x), \quad s \in [0,t].$$

By definition of the operator $\Phi_s^{t,x}$ and expanding the convolutions on the right-hand side above, this equality is equivalent to

$$\langle \Gamma(t-s,x-\star)Z(s),\varphi\rangle_U = \int_{\mathbb{R}^d} \Gamma(t-s,dz)Z(s,x-z) \int_{\mathbb{R}^d} \Lambda(dy) \varphi(x-z-y).$$

Notice that this is precisely the statement of Lemma 4.11 below. Therefore, we can conclude that (4.31) holds.

Step 4. Let us finally sketch the extension of what we have proved so far to the case of Eqs. (4.27) and (4.21). That is, we consider a general Lipschitz continuous drift *b* and initial conditions u_0, v_0 satisfying the hypotheses specified at the beginning of the section. Hence, $\{u(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ satisfies (4.27).

One proves that the process $\{u(t), t \in [0, T]\}$ belongs to $L^2(\Omega \times [0, T]; L^2_{\vartheta})$ as we have done in Step 1. Indeed, an immediate consequence of the proof of Theorem 4.3 is that each term in Eq. (4.27) is bounded in $L^2(\Omega)$, uniformly with respect to $(t, x) \in [0, T] \times \mathbb{R}^d$. This clearly implies that each term in (4.27) defines an element in $L^2(\Omega \times [0, T]; L^2_{\vartheta})$.

It follows that the stochastic integral $\int_0^t \Gamma(t-s) * \sigma(u(s)) dW^*$ is well-defined and, by Steps 2 and 3 above, we have

$$\int_0^t \Gamma(t-s) * \sigma(u(s)) \, d\mathcal{W}^* = \int_0^t \int_{\mathbb{R}^d} \Gamma(t-s, \star -y) \sigma(u(s,y)) \, W(ds, dy),$$

where the \star symbol on the right-hand side stands for the variable in L_{α}^2 .

Concerning the pathwise integral in (4.21), we have

$$\int_0^t \Gamma(t-s) * b(u(s)) \, ds = \int_0^t \, ds \, \int_{\mathbb{R}^d} \Gamma(t-s, dy) \, b(u(s, \star -y))$$
$$= \int_0^t \, ds \, \int_{\mathbb{R}^d} \Gamma(s, dy) \, b(u(t-s, \star -y)).$$

It is also clear that the contributions of the initial conditions in Eqs. (4.27) and (4.21) coincide as elements in $L^2([0, T]; L^2_{\vartheta})$. We have therefore proved that $\{u(t), t \in [0, T]\}$ is the mild solution of (4.21), which concludes the proof of Proposition 4.10. \Box

We now state and prove the following technical lemma, which was used in the proof of Proposition 4.10.

Lemma 4.11. Fix $t \in [0, T]$. Then, for all $\varphi \in \tilde{S}(\mathbb{R}^d)$ and $x \in \mathbb{R}^d$, the stochastic process $\{\Phi_s^{t,x}(\varphi), s \in [0, t]\}$ given by

$$\Phi_{s}^{t,x}(\varphi) = \langle \Gamma(t-s, x-\star)Z(s), \varphi \rangle_{U}$$

coincides, as an element in $L^2(\Omega \times [0, t])$, with $\{\mathcal{K}_s^{t, x}(\varphi), s \in [0, t]\}$, where

$$\mathcal{K}_{s}^{t,x}(\varphi) = \int_{\mathbb{R}^{d}} \Gamma(t-s,dz) Z(s,x-z) \int_{\mathbb{R}^{d}} \Lambda(dy) \, \varphi(x-z-y).$$

Proof. In order to prove the statement, we will first approximate $\{\Phi_s^{t,x}(\varphi), s \in [0, t]\}$ by a sequence of *smooth* processes.

104

More precisely, as it has been explained in [29, Proposition 3.3], for any $(s, x) \in [0, t] \times \mathbb{R}^d$, we can regularize the element $\Gamma(t - s, x - \star)Z(s)$ of U by means of an approximation of the identity $(\psi_n)_n \subset C_0^{\infty}(\mathbb{R}^d)$, and we can assume that ψ_n is symmetric, for all n, and $|\mathcal{F}\psi_n| \leq 1$. Then, for any $s \in [0, t]$, set $J_n^{t,x}(s) := \psi_n * (\Gamma(t - s, x - \star)Z(s))$. Again by [29, Proposition 3.3], $J_n^{t,x}(s)$ belongs to $\mathcal{S}(\mathbb{R}^d)$ and, as $n \to \infty$, $J_n^{t,x}$ converges to $\Gamma(t - \cdot, x - \star)Z(s)$.

$$\Phi_{n,s}^{t,x}(h) := \langle J_n^{t,x}(s), h \rangle_{U}, \qquad h \in U$$

This operator is well-defined because $J_n^{t,x}(s)$ is a smooth function and, in fact, it defines an element in $L^2([0, t] \times \Omega; L_2(U, \mathbb{R}))$.

Moreover, $\Phi_n^{t,x} \to \Phi^{t,x}$ in $L^2([0, t] \times \Omega; L_2(U, \mathbb{R}))$, as $n \to \infty$. Indeed, this is an immediate consequence of the fact that the norm of $\Phi_n^{t,x} - \Phi^{t,x}$ is given by

$$E\left(\int_{0}^{t} \left\| \Phi_{n}^{t,x} - \Phi^{t,x} \right\|_{L_{2}(U,\mathbb{R})}^{2} ds\right) = E\left(\int_{0}^{t} \sum_{j=1}^{\infty} \left| \left\langle J_{n}^{t,x}(s) - \Gamma(t-s,x-\star)Z(s), e_{j} \right\rangle_{U} \right|^{2} ds \right)$$
$$= E\left(\int_{0}^{t} \left\| J_{n}^{t,x}(s) - \Gamma(t-s,x-\star)Z(s) \right\|_{U}^{2} ds \right),$$

where $(e_j)_j$ is a complete orthonormal basis in *U*. The last term above tends to zero because, as mentioned before, $J_n^{t,x} \to \Gamma(t - \cdot, x - \star)Z$ in $L^2([0, t] \times \Omega; U)$.

Therefore, for any $\varphi \in \tilde{S}(\mathbb{R}^d)$ (in fact, for any $\varphi \in U$), the sequence of real-valued processes $(\Phi_n^{t,x}(\varphi))_n$ converges to $\Phi^{t,x}(\varphi)$ in $L^2(\Omega \times [0, t])$. In particular, $\Phi_n^{t,x}(\varphi)$ converges weakly to $\Phi^{t,x}(\varphi)$, that is, for any $0 \le a < b \le t$ and $A \in \mathcal{F}$,

$$E\left(1_{A}\int_{a}^{b}ds \ \Phi_{n,s}^{t,x}(\varphi)\right) \to E\left(1_{A}\int_{a}^{b}ds \ \Phi_{s}^{t,x}(\varphi)\right).$$

$$(4.36)$$

We will conclude the proof by checking that the left-hand side of (4.36) also converges to

$$E\left(1_{A}\int_{a}^{b}ds \ \mathcal{K}_{s}^{t,x}(\varphi)\right). \tag{4.37}$$

For this, note that, by definition of $\Phi_{n,s}^{t,x}$, the left-hand side of (4.36) can be written as

$$E\left(1_A\int_a^b ds \left\langle J_n^{t,x}(s),\varphi\right\rangle_U\right).$$

Because $J_n^{t,x}(s)$ and φ are smooth functions, we can explicitly compute the inner product in the above expression:

$$\begin{split} \left\langle J_n^{t,x}(s),\varphi\right\rangle_U &= \int_{\mathbb{R}^d} \Lambda(dz) \int_{\mathbb{R}^d} dy J_n^{t,x}(s,y)\varphi(y-z) = \int_{\mathbb{R}^d} dy J_n^{t,x}(s,y) \left(\Lambda * \varphi\right)(y) \\ &= \int_{\mathbb{R}^d} dy \left(\int_{\mathbb{R}^d} \Gamma(t-s,dz) \,\psi_n(y-x+z)Z(s,x-z) \right) \left(\Lambda * \varphi\right)(y) \\ &= \int_{\mathbb{R}^d} \Gamma(t-s,dz)Z(s,x-z) \left(\int_{\mathbb{R}^d} dy \,\psi_n(y-x+z) \left(\Lambda * \varphi\right)(y) \right), \end{split}$$

and so the term on the left-hand side of (4.36) equals

$$E\left(1_{A}\int_{a}^{b}ds\int_{\mathbb{R}^{d}}\Gamma(t-s,dz)Z(s,x-z)\int_{\mathbb{R}^{d}}dy\psi_{n}(x-z-y)(\Lambda*\varphi)(y)\right).$$
(4.38)

Since $\varphi \in \tilde{\mathcal{S}}(\mathbb{R}^d)$, the function $y \mapsto (\Lambda * \varphi)(y)$ is continuous in \mathbb{R}^d and $\lim_{|y|\to\infty} (\Lambda * \varphi)(y) = 0$. This implies that, for any $x, z \in \mathbb{R}^d$,

$$\lim_{n \to \infty} \int_{\mathbb{R}^d} dy \, \psi_n(x - z - y) \, (\Lambda * \varphi) \, (y) = (\Lambda * \varphi) \, (x - z)$$

Moreover, because ψ_n and φ belong to $\tilde{S}(\mathbb{R}^d)$, we can apply the definition of the Fourier transform of tempered distributions:

$$\begin{split} \left| \int_{\mathbb{R}^d} dy \, \psi_n(x - z - y) \, (\Lambda * \varphi) \, (y) \right| &= \left| \int_{\mathbb{R}^d} \mu(d\xi) \, \mathcal{F} \psi_n(x - z - \cdot)(\xi) \overline{\mathcal{F} \varphi(\xi)} \right| \\ &\leq \int_{\mathbb{R}^d} \mu(d\xi) |\mathcal{F} \varphi(\xi)| \\ &< +\infty. \end{split}$$

Thus, in order to apply the Dominated Convergence Theorem in (4.38), it remains to prove that

$$E\left(1_A\int_a^b ds \int_{\mathbb{R}^d} \Gamma(t-s,dz)|Z(s,x-z)|\right) < +\infty,$$

and this follows from the hypothesis on Γ and the process *Z*. So we have proved that the limit of (4.38), as *n* goes to infinity, is

$$E\left(1_A\int_a^b ds \int_{\mathbb{R}^d} \Gamma(t-s,dz) Z(s,x-z) \int_{\mathbb{R}^d} \Lambda(dy) \varphi(x-z-y)\right).$$

This shows that the left-hand side of (4.36) converges to (4.37), which concludes the proof.

4.6. Relation with the Dalang–Mueller formulation

In this section, we examine the relationship between the mild random field solution to equation (4.27) and the solution introduced by Dalang and Mueller in [9] (see also [11]), which is based on the L^2 -valued stochastic integration framework that was summarized in Section 2.5. Let L^2_{θ} be the space defined in Section 2.5.

In [9], the authors consider solutions to the following stochastic wave equation in \mathbb{R}^d , for any $d \ge 1$:

$$\frac{\partial^2 u}{\partial t^2}(t,x) - \Delta u(t,x) = \sigma(u(t,x))\dot{W}(t,x), \tag{4.39}$$

with initial conditions

$$u(0, x) = u_0(x), \quad \frac{\partial u}{\partial t}(0, x) = v_0(x), \quad x \in \mathbb{R}^d,$$

where $u_0, v_0 : \mathbb{R}^d \to \mathbb{R}$ are appropriate Borel functions. The noise $\dot{W}(t, x)$ corresponds to the spatially homogeneous Gaussian noise described in Section 2.2.

We denote by $H^{-1}(\mathbb{R}^d)$ the Sobolev space of distributions such that

$$\|v\|_{H^{-1}(\mathbb{R}^d)}^2 := \int_{\mathbb{R}^d} d\xi \, \frac{1}{1+|\xi|^2} |\mathcal{F}v(\xi)|^2 < +\infty.$$

According to [9, Section 5], an adapted L^2_{θ} -valued process { $u(t, \star), t \in [0, T]$ } is a mild L^2_{θ} -valued solution to (4.39) if $t \mapsto u(t, \star)$ is mean-square continuous from [0, T] into L^2_{θ} and the following L^2_{θ} -valued

stochastic integral equation is satisfied:

$$u(t,\star) = \Gamma(t) * v_0 + \frac{\partial}{\partial t} \left(\Gamma(t) * u_0 \right) + \int_0^t \int_{\mathbb{R}^d} \Gamma(t-s,\star-y) \sigma(u(s,y)) M(ds,dy),$$
(4.40)

where Γ denotes the fundamental solution of the wave equation in \mathbb{R}^d . The stochastic integral in (4.40) takes values in L^2_{θ} and is defined in the final part of Section 2.5. The main result on existence and uniqueness of solutions to Eq. (4.40) is the following (see [9, Theorem 13]).

Theorem 4.12. Assume that the spectral measure μ satisfies (4.7), $u_0 \in L^2(\mathbb{R}^d)$, $v_0 \in H^{-1}(\mathbb{R}^d)$ and σ is a Lipschitz function. Then Eq. (4.40) has a unique mild L^2_{θ} -valued solution.

In order to be able to compare the solution of the above equation with the mild random field solution to (4.27), we consider space dimensions $d \in \{1, 2, 3\}$ and we set b = 0. The main result of this section is the following.

Theorem 4.13. Let $d \in \{1, 2, 3\}$, and let $\{u(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ be the mild random field solution of (4.27) in the case of the stochastic wave equation $(d \in \{1, 2, 3\}$ and with b = 0). Let $u(t) = u(t, \star)$. Then $\{u(t), t \in [0, T]\}$ is the L^2_{θ} -valued solution of (4.40).

Proof. For simplicity, we assume that the initial conditions vanish (the extension to the general case is straightforward). Recall that $d \in \{1, 2, 3\}$ and $\{u(t, x), (t, x) \in [0, T] \times \mathbb{R}^d\}$ satisfies the integral equation

$$u(t,x) = \mathcal{I}_{\Gamma,Z}(t,x), \quad a.s. \tag{4.41}$$

for all $(t, x) \in [0, T] \times \mathbb{R}^d$, where

$$\mathcal{I}_{\Gamma,Z}(t,x) := \int_0^t \int_{\mathbb{R}^d} \Gamma(t-s,x-y) Z(s,y) W(ds,dy)$$

and $Z(s, y) := \sigma(u(s, y))$. In order to prove that $\{u(t, \star), t \in [0, T]\}$ is the solution of (4.40), we observe that $u(t, \star) \in L^2_{\theta}$ a.s., since

$$E(\|u(t,\star)\|_{L^2_{\theta}}^2) = \int_{\mathbb{R}^d} E(u(t,x)^2) \,\theta(x) \, dx < \infty$$

by (4.28). Next, we note that $t \mapsto u(t, \star)$ from [0, T] into L^2_{θ} is mean-square continuous, since

$$E(\|u(t,\star) - u(s,\star)\|_{L^2_{\theta}}^2) = \int_{\mathbb{R}^d} E((u(t,x) - u(s,x))^2) \,\theta(x) \, dx,$$

and we observe that as $s \to t$, by (4.28) and since $(t, x) \mapsto u(t, x)$ is $L^2(\Omega)$ -continuous, the right-hand side converges to 0 by the Dominated Convergence Theorem.

For $t \in [0, T]$, define

$$v_{\Gamma,Z}^{\theta}(t,\star) = \int_0^t \int_{\mathbb{R}^d} \Gamma(t-s,\star-y)\sigma(u(s,y)) M(ds,dy),$$

where the stochastic integral is defined as in (2.21). It remains to show that

$$\mathcal{I}_{\Gamma,Z}(t,\star) = v^{\theta}_{\Gamma,Z}(t,\star) \quad \text{in } L^{2}(\Omega \times \mathbb{R}^{d}, dP \times \theta(x)dx).$$
(4.42)

For this, set $Z_n(s, y) = Z(s, y) \mathbb{1}_{[-n, n]^d}(y)$, so that, by definition,

$$v_{\Gamma,Z}^{\theta}(t,\star) = \lim_{n \to \infty} v_{\Gamma,Z_n}^{\theta}(t,\star) \quad \text{in} \quad L^2(\Omega \times \mathbb{R}^d, dP \times \theta(x)dx),$$

where $v_{\Gamma,Z_n}^{\theta}(t,\star)$ is defined as in (2.15). By Proposition 2.11, $v_{\Gamma,Z_n}^{\theta}(t,\star) = \mathcal{I}_{\Gamma,Z_n}(t,\star)$ in $L^2(\Omega \times \mathbb{R}^d, dP \times dx)$, therefore also in $L^2(\Omega \times \mathbb{R}^d, dP \times \theta(x)dx)$. In order to establish (4.42), it suffices to show that

$$E(\|\mathcal{I}_{\Gamma,Z}(t,\star) - \mathcal{I}_{\Gamma,Z_n}(t,\star)\|_{L^2_{\theta}}^2) \to 0 \quad \text{as } n \to \infty.$$

$$(4.43)$$

The expectation in (4.43) is equal to

$$\int_{\mathbb{R}^d} dx \,\theta(x) \, E\left(\left(\mathcal{I}_{\Gamma, Z_n - Z}(t, x)\right)^2\right)$$
$$= \int_{\mathbb{R}^d} dx \,\theta(x) \, E\left[\left(\int_0^t \int_{\mathbb{R}^d} \Gamma(t - s, x - y) \left(Z_n(s, y) - Z(s, y)\right) W(ds, dy)\right)^2\right]$$

For $n \ge t$, let $D_{t,n} = [-n + t, n - t]^d$. As can be seen from the formulas given in Section 4.2, for $x \in D_{t,n}$, the support of $\Gamma(t - s, x - *)$ is contained in $[-n, n]^d$, and by definition, $Z_n(s, y) = Z(s, y)$ for $y \in [-n, n]^d$ n^d and $s \in [0, t]$. Therefore, the above expression is equal to

$$\int_{\mathbb{R}^{d}\setminus D_{t,n}} dx \,\theta(x) \, E\left(\left(\mathcal{I}_{\Gamma,Z_{n-Z}}(t,x)\right)^{2}\right).$$
(4.44)

We notice that by Proposition 2.9,

$$E\left(\left(\mathcal{I}_{\Gamma,Z_n-Z}(t,x)\right)^2\right) \le C\int_0^t ds \sup_{y\in\mathbb{R}^d} E\left(\left(Z_n(s,y)-Z(s,y)\right)^2\right)\int_{\mathbb{R}^d} \mu(d\xi) |\mathcal{F}\Gamma(t-s)(\xi)|^2$$

and

$$\sup_{y\in\mathbb{R}^d} E\left((Z_n(s,y)-Z(s,y))^2\right) \leq \sup_{y\in\mathbb{R}^d} E((Z(t,x))^2) < \infty,$$

by (4.28) and the Lipschitz property of σ . Therefore, the expression in (4.44) converges to 0 as $n \to \infty$. This proves (4.43), and concludes the proof.

Acknowledgement

The authors thank Marta Sanz-Solé for several useful discussions and a careful reading of the manuscript.

References

- [1] J.-P. Aubin, Applied Functional Analysis, John Wiley & Sons, New York-Chichester-Brisbane, 1979.
- [2] Z. Brzeźniak, J. van Neerven, Stochastic convolution in separable Banach spaces and the stochastic linear Cauchy problem, Studia Math. 143 (1) (2000) 43-74.
- [3] R. Carmona, D. Nualart, Random nonlinear wave equations: smoothness of the solutions, Probab. Theory Relat. Fields 79 (4) (1988) 469-508.
- [4] D. Conus, The non-linear stochastic wave equation in high dimensions: existence, Hölder-continuity and Itô-Taylor expansion. Thèse no. 4265, Ecole Polytechnique Fédérale de Lausanne (2008).
- [5] D. Conus, R.C. Dalang, The non-linear stochastic wave equation in high dimensions, Electron. J. Probab. 13 (22) (2008) 629-670.
- [6] R.C. Dalang, Extending the martingale measure stochastic integral with applications to spatially homogeneous s.p.d.e.'s. Electr. J. Probab. 4, Paper No. 6, 29 p. (1999).
- R.C. Dalang, N.E. Frangos, The stochastic wave equation in two spatial dimensions, Ann. Probab. 26 (1) (1998) 187-212.
- [8] R.C. Dalang, O. Lévêque, Second-order hyperbolic s.p.d.e.'s driven by homogeneous Gaussian noise on a hyperplane, Trans. Amer. Math. Soc. 358 (5) (2006) 2123-2159.
- [9] R.C. Dalang, C. Mueller, Some non-linear s.p.d.e.'s that are second order in time, Electron. J. Probab. 8 (1) (2003) 21 pp.
 [10] R.C. Dalang, C. Mueller, L. Zambotti, Hitting properties of parabolic s.p.d.e.'s with reflection, Ann. Probab. 34 (2006) 1423-1450.
- [11] R.C. Dalang, M. Sanz-Solé, Regularity of the sample paths of a class of second-order spde's, J. Funct. Anal. 227 (2) (2005) 304-337.
- [12] R.C. Dalang, M. Sanz-Solé, Hölder-Sobolev regularity of the solution to the stochastic wave equation in dimension 3, Memoirs Amer. Math. Soc. 199 (931) (2009).

- [13] D. Dawson, Stochastic evolution equations, Math. Biosci. 154 (1972) 287-316.
- [14] G. Da Prato, J. Zabczyk, Stochastic Equations in Infinite Dimensions, Cambridge University Press, Cambridge, 1992.
- [15] G. Da Prato, J. Zabczyk, Ergodicity for Infinite-dimensional Systems. London Mathematical Society Lecture Note Series, 229, Cambridge University Press, Cambridge, 1996.
- [16] L.C. Evans, Partial Differential Equations, American Mathematical Society, Providence, RI, 1998.
- [17] G.B. Folland, Introduction to Partial Differential Equations, Princeton Univ. Press, 1976.
- [18] A. Grorud, E. Pardoux, Intégrales hilbertiennes anticipantes par rapport à un processus de Wiener cylindrique et calcul stochastique associé, Appl. Math. Optim. 25 (1) (1992) 31-49.
- [19] A. Karczewska, Stochastic integral with respect to cylindrical Wiener process, Ann. Univ. Mariae Curie-Sklodowska Sect. A 52 (2) (1998) 79–93.
- [20] A. Karczewska, J. Zabczyk, A note on stochastic wave equations. Evolution equations and their applications in physical and life sciences (Bad Herrenalb, 1998), 501–511, Lecture Notes in Pure and Appl. Math. 215, Dekker, New York, 2001.
- [21] A. Karczewska, J. Zabczyk, Stochastic PDEs with function-valued solutions. Infinite dimensional stochastic analysis (Amsterdam, 1999), 197–216, Verh. Afd. Natuurkd. 1. Reeks. K. Ned. Akad. Wet. 52, R. Neth. Acad. Arts Sci., Amsterdam, 2000.
- [22] N.V. Krylov, An analytic approach to SPDEs. Stochastic partial differential equations: six perspectives, 185–242, Math. Surveys Monogr. 64, Amer. Math. Soc., Providence, RI, 1999.
- [23] N.V. Krylov, B.L. Rozovskii, Stochastic evolution equations. (Russian) Current Problems in Mathematics, Vol. 14 (Russian), 71–147, 256, Akad. Nauk SSSR, Moscow, 1979.
- [24] D. Márquez-Carreras, M. Mellouk, M. Sarrà, On stochastic partial differential equations with spatially correlated noise: smoothness of the law, Stoch. Proc. Appl. 93 (2001) 269–284.
- [25] M. Métivier, J. Pellaumail, Stochastic Integration. Probability and Mathematical Statistics, Academic Press, New York-London-Toronto, 1980.
- [26] A. Millet, M. Sanz-Solé, A stochastic wave equation in two space dimensions: smoothness of the law, Ann. Probab. 27 (2) (1999) 803–844.
- [27] A. Millet, M. Sanz-Solé, Approximation and support theorem for a wave equation in two space dimensions, Bernoulli 6 (5) (2000) 887–915.
- [28] C. Mueller, Long time existence for the wave equation with a noise term, Ann. Probab. 25 (1997) 133-151.
- [29] D. Nualart, L. Quer-Sardanyons, Existence and smoothness of the density for spatially homogeneous spde's, Potential Analysis 27 (3) (2007) 281–299.
- [30] S. Peszat, The Cauchy problem for a nonlinear stochastic wave equation in any dimension, J. Evol. Equ. 2(3)(2002) 383–394.
- [31] S. Peszat, J. Zabczyk, Stochastic evolution equations with a spatially homogeneous Wiener process, Stochastic Process. Appl. 72 (2) (1997) 187-204.
- [32] S. Peszat, J. Zabczyk, Nonlinear stochastic wave and heat equations, Probab. Theory Related Fields 116 (3) (2000) 421–443.
- [33] C. Prévôt, M. Röckner, A Concise Course on Stochastic Partial Differential Equations. Lecture Notes in Mathematics, 1905, Springer, Berlin, 2007.
- [34] L. Quer-Sardanyons, The stochastic wave equation: study of the law and approximations. PhD Thesis, Universitat de Barcelona, 2005.
- [35] L. Quer-Sardanyons, M. Sanz-Solé, Absolute continuity of the law of the solution to the 3-dimensional stochastic wave equation, J. Funct. Anal. 206 (1) (2004) 1–32.
- [36] L. Quer-Sardanyons, M. Sanz-Solé, A stochastic wave equation in dimension 3: smoothness of the law, Bernoulli 10 (1) (2004) 165–186.
- [37] B.L. Rozovskii, Stochastic evolution systems. Linear theory and applications to nonlinear filtering. Translated from the Russian by A. Yarkho. Mathematics and its Applications (Soviet Series), vol. 35, Kluwer Academic Publishers Group, Dordrecht, 1990.
- [38] M. Sanz-Solé, Malliavin calculus, with applications to stochastic partial differential equations. Fundamental Sciences, EPFL Press, Lausanne; distributed by CRC Press, Boca Raton, FL, 2005.
- [39] L. Schwartz, Théorie des Distributions, Hermann, Paris, 1966.
- [40] J. van Neerven, L. Weis, Stochastic integration of functions with values in a Banach space, Studia Math. 166 (2) (2005) 131–170.
- [41] J.B. Walsh, An introduction to stochastic partial differential equations. Ecole d'Eté de Probabilités de Saint-Flour XIV 1984, Lect. Notes Math. 1180 (1986) 265–437.
- [42] J. Weidmann, Linear operators in Hilbert spaces. Graduate Texts in Mathematics, 68, Springer-Verlag, New York-Berlin, 1980.